Automating Fake News Detection System using Multi-level Voting Model

Sawinder Kaur $\,\cdot\,$ Parteek Kumar $\,\cdot\,$ Ponnurangam Kumaraguru

Received: date / Accepted: date

Abstract The issues of online fake news have attained an increasing eminence in the diffusion of shaping news stories online. Misleading or unreliable information in form of videos, posts, articles, URLs are extensively disseminated through popular social media platforms such as Facebook, Twitter, etc. As a result, editors and journalists are in need of new tools that can help them to pace up the verification process for the content that has been originated from social media. Motivated by the need for automated detection of fake news, the goal is to find out which classification model identifies phony features accurately using three feature extraction techniques, Term Frequency-Inverse Document Frequency (Tf-Idf), Count-Vectorizer (CV) and Hashing-Vectorizer (HV). Also, in this paper, a novel multi-level voting ensemble model is proposed. The proposed system has been tested on three datasets using twelve classifiers. These ML classifiers are combined based on their false prediction ratio. It has been observed that the Passive Aggressive (PA), Logistic Regression (LR) and Linear Support Vector (LinearSVC) individually performs best using TF-IDF, CV and HV feature extraction approaches, respectively based on their

Sawinder Kaur (Corresponding Author) Doctoral Research Lab-II, Computer Science and Engineering Department, TIET, Patiala, India. Tel.: +91-8837761601 E-mail: sawinderkaurvohra@gmail.com Parteek Kumar

Computer Science and Engineering Department, TIET, Patiala, India. E-mail: parteek.bhatia@thapar.edu

Ponnurangam Kumaraguru Computer Science and Engineering Department, IIIT, Delhi, India. E-mail: pk@iiitd.ac.in performance metrics. Whereas, the proposed model outperforms the Passive Aggressive model by 0.8%, Logistic Regression model by 1.3%, LinearSVC model by 0.4% using TF-IDF, CV and HV, respectively. The proposed system can also be used to predict the fake content (textual form) from online social media websites.

Keywords Fake news articles, Count-Vectorizer, Tf-Idf, Hashing-Vectorizer, Classifiers, Textual content, Machine learning models

1 Introduction

A growing interest related to fake news detection has attracted many researchers as fake information is circulated through online social media platforms such as Facebook, Twitter, etc. The fake content is spreading at a faster pace to gain popularity over social media, to distract people from the current critical issues. Most of the people believe that the information they receive from various social media sites is reliable and true, *i.e.*, people are inherently truth-biased. Also, people easily trust and want to believe in what they actually interpret in their minds, *i.e.*, confirmation-biased. In general, it has been analyzed that humans are unable to recognize deception effectively. Due to which a serious and negative impact of fake articles can be seen on society and individuals leading to an imbalance of the news ecosystem. It was observed that during US president election [1], most of the widely spread articles on social media were fake. Recently a fake video related to Kerela battling with floods was viral on social media platform (Facebook) claiming that the Chief Minister of the state is forcing the Indian Army to stop conducting the rescue operations in flooded regions of Kerela. Also,

during India's national election (2019), various WhatsApp groups (> 900,000) were created to disseminate the fake information regarding India's ruling party [2]. Most of the fake articles are created to confuse people and trigger their distrust. Such problems led researchers to look at some automated ways to access the ground truth values of fake text on the basis of the textual content posted in articles on social platforms.

Social media enables to maintain and develop relations with others. It helps the users to present themselves by creating their profiles, sharing information through photos, images, and text to link with other members [3]. Some of the most popular social media [4] sites are Facebook, Twitter [5] [6] [7], Instagram [8], WhatsApp [9] [10], LinkedIn, WeChat, Snapchat, Foursquare [11]. With the popularity of social media sites [12], level of usage to share content on online social media has increased. There are several reasons for the change in behavior for such kind of consumptions. The content shared on social media platforms requires less time and cost than on newspapers or traditional news media. Easier to share content in the form of video, blogs, posts with friends or users. This gives the growth ease to the authors and publishers to publish their contents as articles on collaborative environments. There is 13% of the global increase in social media usage since 2017 [9]. Distribution and creation of news content in the form of posts [13], blogs, articles, images, videos, etc., have been spreading through social media websites [12]. This increase in social media also gives rise in the spread of fake articles [14] over the internet.

1.1 Types of Fake News

According to the literature, there are five types of fakes news. The first type can be seen in the form of **deliber**ate misinformation, which is misleading information that is spread in a calculated way to deceive targeted users. Other forms of fake news can be **clickbait** [15] [16] which grab the reader's attention with a purpose to make them click on the fake news seen on the internet. Users who set up fake sites generate huge revenues and clicking on such websites results in bombarded ads. Articles [14] from satirical sources like 'The Onion' often repeat and share the news as if the printed stories were true. Parody or Satirical [17] articles use obscenity, absurdity and exaggeration to comment on current events and to unease the readers. False head**lines** are intentionally exaggerated to draw the reader's attention. In such headlines, the title of the articles may not match the context of stories, the headline can be read as one way and state something different as a fact.

This type of fake news is untrue at worst and misleading at best. **Hoaxes** is another type of misinformation which deceives the reader deliberately by causing harm and material losses to the users.

1.2 Contribution

Researchers have analyzed that an automated system sometimes identifies fake news articles better than human do. The automated systems can play an important tool for identifying fake stories, articles, blogs, clicks which manipulate public opinion on social networking sites [18] [19]. Taking into need for the development of such fake news detection system, in this paper we have identified news articles as fake or real by using supervised machine learning classifiers such as Naïve Bayes (NB), Decision Tree (DT), Support Vector Machine (SVM), Linear models, Neural Networks (NN) and Ensemble models. To get effective results, three different corpora (News Trends, Kaggle and Reuters) have been collected with similar characteristics. The feature extraction techniques, Term Frequency-Inverted Document Frequency (Tf-Idf), Count-Vectorizer (CV), Hashing-Vectorizer (HV) are used to extract feature vectors from the textual content of articles. The effectiveness of a set of classifiers is observed when they predict the labels of the testing samples after learning the training data using these features extraction techniques.

Various supervised ML models are extensively used for categorization of textual data as fake or real, but such models were not able to obtain the results of an ideal classifier. So, a multi-level voting model has been proposed in this paper to build an ideal classifier with significant improvement than previously existing models. Our contribution to this paper is as follows.

- Statistical analysis of collected datasets (News Trends, Kaggle and Reuters) with negative and positive instances has been performed.
- Twelve ML models are evaluated using Tf-Idf, CV, HV feature extraction techniques to retrieve the best model based on performance metrics.
- A novel method is proposed to merge the ML models based on false prediction rate.
- Proposed an ideal multi-level voting classifier (threelevel) to verify the effectiveness of the ensemble model.
- A comparative study is performed to show the effectiveness of our proposed model.

The performance of the multi-level voting system is analyzed using parameters like precision, recall, specificity, ROC curve, F1-score. The remaining paper is organized as Section 2 gives a brief overview of the related work done in the field of fake news classification. The problem statement is discussed in Section 3. Section 4 covers the methodology of the proposed system. Section 5 introduces the classifiers used for detecting the fake and real articles. The evaluation phase along with analysis done by all classifiers is presented in Section 6. The proposed model is discussed in Section 7 and its performance is evaluated in Section 8. The comparison of the proposed model with existing work is discussed in Section 9. Section 10 concludes the paper along with its future works.

2 Related Work

Many techniques have been developed recently for fake news detection. In this section, the closely related research work that has been done on detecting fake news on online social media is discussed.

Most of the social sites require energy and time to manually remove or filter spam. Markines et al. (2009) proposed six highlights (TagSpam, TagBlur, DomFp, NumAds, Plagiarism, ValidLinks) of tagging systems catching diverse properties of social spam [20]. Utilizing the six proposed highlights, creators assessed different administered machine learning techniques to identify spam with precision over 98% with a false positive rate of 2%. The Weka tool used for the experiment gives the best accuracy when evaluated on the basis of Adaboost classifier. To address the issue of detecting video promoters and spammers, Benevenuto et al. (2009) manually assembled test gathering of genuine YouTube clients and classified them as legitimates, spammers and promoters. Authors have investigated the feasibility for detecting spammers and promoters by applying a supervised classification algorithm [21]. The classifier used in this paper correctly identifies the majority of promoters correctly.

Qazvinian *et al.* (2011) explored three features such as content-based, network-based and microblog specific memes to correctly identify the rumors [22]. Such features are also used to identify disinformers or users who endorse a rumor and further tries to spread it. For the experiment, authors collected 10,000 manually annotated tweets from twitter and achieved 0.95 in Mean Average Precision (MAP). Rubin *et al.* (2016) proposed a satire detection model with Support Vector Machine (SVM) based algorithm across 4 domains, such as science, business, soft news and civics [17]. To verify the sources of news articles, authors have discussed various legitimate and satirical news websites. In this paper, five features together are chosen to predict the best predicting feature combination with 90% precision and 84% recall to identify satirical news which can help to minimize deception impact of satire.

Analysis for the real event such as Boston Marathon Blasts is done by Gupta *et al.* (2013). During the event, it was observed that a lot of fake and malicious profiles originated on Twitter. Results showed that 29% of the content originated during the Boston Blasts [5] was viral whereas 51% was generic opinions and comments. Authors identified six thousand profiles which were immediately created after the blasts occurred and were suspended by Twitter. Tabloids are often used for sensationalization, exaggeration, producing misleading and low-quality news content. A new form of tabloidization has emerged known as clickbaiting. There exists both non-textual and textual clickbaiting, which is surveyed by Chen *et al.* (2015), who proposed a hybrid approach [15] for automatic detection of clickbait.

Rubin *et al.* (2015) utilized Vector Space Modelling (VSM) and Rhetorical Structure Theory (RST) to analyze misleading and truthful news. RST catches the coherence of story in terms of functional relations among the useful text units and also describe the hierarchical structure for each article/news. VSM is used for identifying the relations among rhetorical structure [23], *i.e.*, each article content can be depicted as vectors in high dimensional space.

Researchers have used different techniques to identify and review the fake content. One of the best and common feature extraction method is Bag of Words. This comprises of a group of words retrieved from the textual content, from where n-gram [4] features can be extracted. The second most important feature which is similar to the Bag of Words approach is Term Frequency (TF) which is related to frequency of the words. Conroy *et al.* (2015) proposed a hybrid approach which combines both machine learning and linguistic cues with network-based behavioral data [1]. The hybrid approach follows both n-gram and bag of word techniques to represent data. Ahmed et al. (2017) proposed a fake news detection system which uses n-gram [24] analysis and Term Frequency- Inverse Document Frequency (Tf-Idf) as a feature extraction technique [4]. In this paper, six classifiers of machine learning are used and two different feature extraction techniques are used for comparison and investigation. Volkova et al. (2017) built a predictive model to manage 130K news posts as verified or malicious. Authors have classified four subtypes of suspicious news such as propaganda, clickbait, hoaxes and satire [25].

Chhabra *et al.* (2011) had put forward a URL static feature based detection method for detecting malicious websites with accurate results. The author has focussed on external features such as IP addresses. Further, a vector construction VSM [26] is chosen as the URL vector model. The dataset taken in this paper consists of malicious URLs which was downloaded from the phishing platform named as 'Phishtank'[27].

In our digital world, fake news is disseminating and impacting millions of users on social media platforms every day. It has really become difficult to separate truth from fiction. With the help of machine learning models, it is possible to detect spam emails at an early stage with the help of spam filters. ML classifiers help to solve the real world problems. Also, ML has made easier for the users of e-commerce business as it helps to identify the hidden pattern, groups the similar products into a cluster and displays the result to end-user which enables a product based recommendation system. It also helps to solve the problem of unfair recommendations [28].

Comparative analysis of related work done in the field of fake news detection and the proposed system presented in this paper is shown in Table 1.

It has been analyzed that the research work done in the field of fake news detection is mainly restricted to SVM and Naïve Bayes classifiers using only n-gram [24] and Tf-Idf features extraction approaches. No work has been done on Multi-Layer Perceptron (MLP), Long Short term Memory (LSTM) [29] models and hashing based extraction approach which also accounts for better efficiency. Also, the Existing fake news detection [30] models are built using supervised machine learning algorithms whereas manual hand-crafted feature extraction is more time consuming and inefficient method to achieve the best accuracy. Existing techniques studied so far provides a direction to be followed further for quantitative and qualitative research.

3 Problem Statement

To address the issue of fake news generation and dissemination through various online social platforms, an appropriate feature extraction technique is chosen to improve the efficiency of the existing ML classifiers. A novel multi-level voting ensemble model will be proposed to develop an efficient fake news detection system. Mathematically, the problem statement can be represented as- To identify $S = \{fake, real\}$ for a document D where $D = \{t_1, t_2, ..., t_n\}$ and t_i represents the text in a news article a_i chosen from corpus with series of engagements that is composed of title, body and label of the article as $e_{ijk} = (t_i, b_j, l_k)$. The task is to evaluate and analyze the best feature extraction method f_m where $m = \{\text{Tf-Idf, CV, HV}\}$ using machine learning classifier to compute high efficiency in our proposed system. The approach followed in this paper has been discussed in the next Section.

4 Methodology

The architecture of the proposed fake news article detection system [31] is seen in Figure 1. To train the system three corpora has been collected from three different sources by downloading the datasets from News Trends, Kaggle and Reuters websites. In the pre-processing phase, stop words and duplicate text from news articles is removed. The missing values *i.e.*, not available (NA) values are collected and cleaned in the next step. The data retrieved is then split into two parts, training (0.67) and testing (0.33) sets. The feature extraction phase is then carried out to retrieve meaningful features from the textual data. In this phase, the features are extracted from the articles. Three feature extraction techniques such as Term Frequency-Inverted Document Frequency (assigns weights according to the importance of the terms in the document), Count-Vectorizer (counts the frequency of the terms in a document) and Hashing-Vectorizer (follows the hashing trick) have been applied. The features retrieved are then fed to the classification algorithm chosen in next phase. The various ML models such as MultinomialNB, Passive Aggressive, Stochastic Gradient Descent, Logistic Regression, Support Vector Classifier, Nu-Support Vector Classifier, Multi-Layer Perceptron, Linear-Support Vector Classifier, AdaBoost, Gradient Boosting, Decision Tree, Voting classifiers [32] are chosen to learn and identify the patterns and outcomes from them. The models are then evaluated based on performance metrics to achieve an efficient classifier. Based on the analysis done, the models are integrated to propose a multi-level voting model to achieve high efficiency and then is compared with the existing work [33] as discussed in Section 9. The detailed working of each phase, that has been implemented in python framework is discussed below.

4.1 Data Collection

There are many sources of fake article generation such as Facebook [34], Twitter [35] [36] [37] [38], etc., which are used as a trading platform to disseminate fake news. We have used News Trends [39], Kaggle [40] and Reuters [41] dataset with similar attributes such as headlines, body, publisher name of the article, published date, categorical and missing values. The News Trends, Kaggle, and Reuters corpus consist of 7,795; 4,048 and 21578 news articles labeled as fake and real news, respectively.

JaminolsLinguisticJaminolsJaminolsJaminolsJaminolsJaminolsJaminolsJaminolsJaminolsat 100000distinct six features for et al. (2009)SVM, AdaBoostSvM, AdaBoostTagSpam, TagBlur, Tagspam, TagBlur, DomFp, NumAds, Plagia- rism, ValidLinksBenevenuto et al. (2009)A video response crawler is pro- posed to identify spammers in online video social network.SVMReal YouTube user infor- mation.Video attributes, individ- ual characteristics of user behavior, social relation between users via video re- sponse interactions.Qazinian et al. (2011)Identified tweets in which rumor is endorsed.Naïve Bayes Logistic Re- gression, DT, SVM-BF, SVM-BF, SVM-SigmoidTweetsContent-based, network- based, Twitter specific memes.Gupta et al. (2013)Analysis of Twitter content duri news.Logistic Regres- sionTweets and correspond- ing user informationTopic engagement, Global engagement, Global engagement, Global engagement, Global mems.Topic engagement, Global engagement, Global engagement, Global engagement, Global engagement, Global informationChen et al. (2015)Analyzed coherence relations be tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'Rubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.SVMUS and Canadian national newspapersAbsurdity, Humor, Gram- mar, Negative affect, Punctuation.Ahmed et al. (2015)Developed n	Internord Import of proper and spanners using machine learning. SWM, AdaBoost Spanne posts, tags. TagSpan, TagB Benevenuto et al. (2009) A video response crawler is prosol identify spanners in online video social network. SVM Real Video attributes, individentiations. Qazinian et al. (2011) Identified tweets in which rumor is endorsed. Naïve Bayes. Tweets Content-based, network based, Twitter spectra attributes, individentiations. Qazinian et al. (2011) Using URLs static features, a al. (2011) Naïve Bayes. Malicious Granmar, Lexical, Varter spectra attributes, individentiations. Gupta et al. (2011) Malyzed cherence relations between users via video social retation. Naïve Bayes. Malicious Granmar, Lexical, Varter spectra attributes, individentiations. Gupta et al. (2013) Analyzed coherence relations between deceptive and truthful news. Naïve Bayes. Malicious Granmar, Lexical, Varter spectra attributes, individentiation. (2015) Analyzed coherence relations between deceptive and truthful news. VSM News same ples from NPR's 'Bhiff Topic engagement, Gle canadian national newspapers (2015) A spirite detection model is developed no posed combining linguistic and network-based behavior data. SVM News same ples from NPR's 'Bhiff Discourse (2015)	Authors	Proposed Approach		Dataset	Features
Markings etAnalysed discher sin earlies for machine learning.SVA, AnalossSpan poss, tag.Tag.nin, tag.Tag.nin, rism, ValidLinksBenevenuto et al. (2009)A video response crawler is pro- posed to identify spammers in online video social network.SVMReal YouTube user infor- nation.Wideo attributes, individ- ual characteristics of user behavior, social relation between users via video re- sponse interactions.Qazinian et al. (2011)Identified tweets in which rumor is endorsed.Naïve Bayes Logistic Re- gression, DT, SVM-RBF, SVM-Linear, SVM-SigmoidMairee Bayes, Logistic Re- gression, DT, SVM-Linear, SVM-SigmoidMairee Bayes, Logistic Re- gression, DT, SVM-RBF, SVM-Linear, SVM-SigmoidMairee Bayes, Content-based, network- based, Twitter specific memes.Gupta et al. (2013)Analyzed coherence relations be- tween deceptive and truthful news.Logistic Regres- sionTweets and correspond- ing user, Social repu- tain, Likability, Credi- bilityChen et al. (2015)A hybrid approach is pro- oposed combining linguistic and network-based behavior data.VSMNews sam- ples from NPR's Bluff the Listener'Rubin et al. (2015)A hybrid approach is pro- oped.SVMUS and Canadian national network-based classi-Absurdity, Humor, Gram- mar, Negative affect, Punctuation.(2015)Developed n-gram based classi- fer to differentiate between fake ad real articles.SVMUS and Canadian national newspapersAbsurdity, Humor, Gram- mar, Negative affect, Punctuation.<	All (2009) A video response crawler is pro- ord al. (2009) SVM, Auaboust of Yar, Auaboust of Yaretrifecres and transer of Yar, Auaboust of Yar, Auabo	Markinos et	Analyzed distinct six features for	SVM AdaBoost	Spam posts	TagSpam TagBlur
al. (2007) detecting social spanners using machine learning. rism, ValidLinks Benevenuto et al. (2009) A video response crawler is proposed to identify spanners in online video social network. SVM Real YouTube users via video response information. Video attributes, individual characteristics of user sponse interactions. Qazinian et al. (2011) Identified tweets in which rumor is endorsed. Naïve Bayes. Tweets Content-based, network-based, network-based, restrict, Vec-tors and Static. Chhabra et al. (2011) Using URLs static features, a method is developed to detect malicious websites. Naïve Bayes, SVM-RBF, tank' Malicious Grammar, Lexical, Vec-tors and Static. SVM-RBF, tank' SVM-Sigmoid Tweets and correspond-ing user Topic engagement, Global engagement, Social reputing user (2013) Ing Boston Marathon. Logistic Regression, DT, SVM-RBF, tank' Tweets and correspond-ing user Topic engagement, Global engagement, Social reputing user (2015) A halyzed coherence relations be tween deceptive and truthful news. VSM News samples from NPR's 'Bluff (2015) A satire detection model is developed oped. SVM US and Canadian national network-based behavior data. Absurdity, Humor, Grammar, Lexialon. (2015) Developed n-gram based classic and real articles. SVM News articles	Int. (2007) Interesting spanners using machine learning. Int. (2008) Interesting spanners in online video social network. Benevenuto et al. (2009) video tresponse crawler is proson online video social network. SVM Real Yourube user information. Qazinian et al. (2011) Identified tweets in which rumor is endorsed. Naïve Bayes Tweets Content-based, network. Chhabra et al. (2011) Using URLs static features, a method is developed to detect malicious websites. Naïve Bayes. Malicious Grammar, Lexical, V tors and Static. SVM-RBF, SVM-RBF, SVM-RBF, (2013) method is developed to detect in g Boston Marathon. Logistic Regression, DT, from "Phish-tank" Topic engagement, Gocial rel ank? Chen et al. (2015) Analyzed coherence relations between deceptive and truthful news. VSM News samples from NPR's 'Bluff Rubin et al. (2015) A hybrid approach is proposed combining linguistic and network-based behavior data. SVM Simple text sentences Bag of Words, n-gram arx, Negative affer Punctuation. (2015) Developed n-gram based classi- classi- differentiate between fake ad real articles. SVM News articles TF-IDF (2017) Entervise of suspicions news; satire, hoaxes, click-bait and propaganda. SGD, PA, News articles TF-IDF, Coc Proposed <td>al (2000)</td> <td>detecting social spammers using</td> <td>SVM, Adaboost</td> <td>tora</td> <td>DomEn NumAda Plagia</td>	al (2000)	detecting social spammers using	SVM, Adaboost	tora	DomEn NumAda Plagia
Benevenuto et al. (2009)A video response crawler is pro- posed to identify spammers in online video social network.SVMReal ware 	Benevenut et al. (2009)A video response crawler is pro- posed to identify spammers in online video social network.SVMReal YouTube user infor- mation.Video attributes, indi- ual characteristics of t behavior, social relat between users via video sponse interactions.Qazinian et al. (2011)Identified tweets in which rumor is endorsed.Naïve Bayes, Logistic Re- UKL dataset gression, DT, SVM-RBF, SVM-RBF, SVM-RBF, SVM-SigmoidTweetsContent-based, network based, network based, network based based of wither content dur- ing Boston Marathon.Naïve Bayes, Logistic Re- gression, DT, SVM-RBF, SVM-SigmoidMalicious ters and Static.Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.Naïve Same sonTopic engagement, Gle engagement, Social re tationChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from work modelsDiscourseRubin et al. (2015)A satire detection model is devel- oped.SVMUS and Absurdity, Humor, Gr mar, Negative affi punctuation.Rubin et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesCanation (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesCatation et (2018)A predictive model was built to predict 4 subtypes of suspi cious news; satire, hoaxes, click- bait and propaganda.SGD, PA, alNB, Gradient alNB, Gradient alNB,	ui. (2009)	machine learning		tags.	pigm Valid inla
Benevenuto A video response cravier is pro- posed to identify spammers in online video social network. SVM Real Video attributes, individ- gravitics of user infor- mation. Video attributes, individ- user infor- sponse interactions. Qazinian et al. (2011) Identified tweets in which rumor is endorsed. Naïve Bayes. Tweets Content-based, network- behavior, social relation between users via video re- sponse interactions. Chhabra et al. (2011) Using URLs static features, a method is developed to detect malicious websites. Naïve Bayes, SVM-RBF, SVM-RBF, SVM-RBF, SVM-RBF, SVM-RBF, SVM-RBF, SVM-Sigmoid Malicious URL dataset from 'Phish- tamk' Topic engagement, Global correspond- ing user information Gupta et al. (2013) Analysis of Twitter content dur- ing Boston Marathon. Logistic Regres- sion Tweets and sion Topic engagement, Global correspond- ing user information Chen et al. (2015) Analyzed coherence relations be- tween deceptive and truthful news. VSM News sam- ples from NPR's 'Bluff Rubin et al. (2015) A hybrid approach is pro- posed combining linguistic and network-based behavior data. SVM US and canadian national newspapers Absurdity, Humor, Gram- mar, Negative affect, Punctuation. Ahmed et al. (2017) Developed n-gram based classi- fier to differentiate between fake ad real articles. Linguistic mod- karticles. News posts TF-IDF, Doc2Vec	Beneventito A video response crawler is pro- posed to identify spammers in online video social network. SVM Real Video attributes, indo- mation. Video attributes, indo- user infor- mation. video attributes, indo- sentareteristics of u behavior, social relat between users via video sponse interactions. Qazinian et al. (2011) Identified tweets in which rumor is endorsed. Naïve Bayes, method is developed to detect malicious websites. Naïve Bayes, method is developed to detect malicious websites. Malicious Logistic Re- gression, DT, SVM-Bigmoid Grammar, Lexical, V tors and Static. Gupta et al. (2013) Analysis of Twitter content dur- ing Boston Marathon. Logistic Regres- information Tweets and correspond- butween deceptive and truthful news. Topic engagement, Glo engagement, Social re tation, Likability, Cr bility Chen et al. (2015) Analyzed coherence relations be- tween deceptive and truthful news. VSM News sam- ples from NPR's 'Bluff Discourse Rubin et al. (2015) A staire detection model is devel- oped. SVM US and Canadian national network-based behavior data. SVM US and Canadian national network-based classi- fier to differentiate between fake ad real articles. SVM News articles TF-IDF, Doc2Vee (2017) A predictive model was built to predict 4 subtypes of supj- cious news; satire, hoaxes, click- bait and propagnda. SGD, PA, alNB, Gradient bosoting, News articles <t< td=""><td>D (</td><td>machine learning.</td><td>CIVI</td><td></td><td></td></t<>	D (machine learning.	CIVI		
et al. (2009) posed to identify spammers in online video social network. You lube ual characteristics of user information. Qazinian et al. (2011) Identified tweets in which rumor is endorsed. Naïve Bayes Tweets Content-based, network-based, Twitter specific memes. Chhabra et al. (2011) Using URLs static features, a al (2011) Naïve Bayes, with the specific memes. Malicious Grammar, Lexical, Vector memes. Control is developed to detect malicious websites. Naïve Bayes, SVM-RBF, SVM-RBF, SVM-RBF, SVM-RBF, SVM-RBF, SVM-RBF, SVM-Sigmoid Malicious Grammar, Lexical, Vectors and Static. Gupta et al. Analysis of Twitter content during Boston Marathon. Logistic Regression Tweets and corresponding user information Topic engagement, Global engagement, Social reputing user information (2013) ing Boston Marathon. VSM News same Discourse (2015) Analyzed coherence relations between deceptive and truthful news. VSM News same Discourse (2015) A hybrid approach is proposed oped. SVM Simple text sentences Bag of Words, n-gram sentences (2015) A satire detection model is developed noged. SVM US and charational national natonal national	et al. (2009) posed to identify spammers in online video social network. You Tube ual characteristics of the user information. Qazinian et al. [dentified tweets in which rumor al. (2011) Identified tweets in which rumor is endorsed. Naïve Bayes. Tweets Content-based, network. Chhabra et al. (2011) Using URLs static features, a method is developed to detect malicious websites. Naïve Bayes. Malicious Grammar, Lexical, V tors and Static. Gupta et al. Analysis of Twitter content during Boston Marathon. Logistic Resting SWM-Linear, SVM-Sigmoid Topic engagement, Glc corresponding user information Topic engagement, Glc corresponding user information (2013) Analyzed coherence relations between deceptive and truthful news. VSM News samples from NPR's 'Bluff Discourse (2015) posed combining linguistic and network-based behavior data. SVM US and Canadian national network-based behavior data. SVM Sumple text senteres Absurdity, Humor, Gramar, Negative affirm national network-based behavior data. Corroy et al. A satire detection model is devel oped. SVM US and Canadian national network-based behavior data. TF-IDF, Doc2Vec Catano et al. (2018) Developed n-gram based classificions news; satire, hoaxes, click-bait and propagnada. SGD, PA, News articles TF-IDF, Correv Vectorizer, Hashi Vectori	Benevenuto	A video response crawler is pro-	SVM	Real	video attributes, individ-
online video social network.user mlor mation.behavior, social relation mation.Qazinian al. (2011)identified tweets in which rumor is endorsed.Naïve Bayes network.TweetsContent-based, network- based, Twitter specific memes.Chhabra al. (2011)et user malicious websites.Using URLs static features, a method is developed to detect malicious websites.Naïve Bayes, Logistic Re- gression, DT, SVM-RBF, SVM-RBF, SVM-SigmoidMalicious URL dataset from 'Phish- tank'Grammar, Lexical, Vec- tors and Static.Gupta (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sionTweets and correspond- informationTopic engagement, Global engagement, Social repu- tation, Likability, Credi- bilityChen (2015)analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- oped.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affect, Punctuation.Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.Linguistic model elsNews postsTF-IDFCatano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news, satire, hoaxes, click- kit end urgensewick, click- kit end urgensewickLinguistic mod- elsNews postsTF-IDF, Doc2Vec	online video social network.user mation.user mation.behavor, social relial behavor, social relial behaver, social relial <td>et al. (2009)</td> <td>posed to identify spammers in</td> <td></td> <td>YouTube</td> <td>ual characteristics of user</td>	et al. (2009)	posed to identify spammers in		YouTube	ual characteristics of user
Qazinian et al. (2011)Identified tweets in which rumor is endorsed.Naïve BayesTweetsContent-based, network- based, Twitter specific memes.Chhabra et 	QazinianetIdentified tweets in which rumor is endorsed.Naïve BayesTweetsContent-based, netwo based, Twitter spec memes.ChhabraetUsing URLs static features, a method is developed to detect malicious websites.Naïve Bayes, Logistic Re- gression, DT, SVM-RBF, SVM-Linear, SVM-SigmoidMalicious URL dataset from 'Phish- tank'Taranmar, Lexical, V tors and Static.GuptaetAnalysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sionTweets and correspond- ing user informationTopic engagement, Gle engagement, Social re tation, Likability, Cr bilityChenetal.Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'BluffRubinetA hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affic portunation.Ahmedet al. (2015)Developed n-gram based classi- for o predict 4 subtypes of suspi- to predict 4 subtypes of suspi- to suite, soits and propaganda.Linguistic mod- elsNews postsTF-IDF, Doc2VecAhmed et al. (2018)Predictive model was built to predict 4 subtypes of suspi- to sing textual data of articles, and stores, click- bait and propaganda.SGD, PA, alNB, Gradient Bosting,News articlesTF-IDF, CorProposed is developed to detect fake arti- is developed to detect fake arti- clos.SGD, PA, alNB, GradientNews articlesTF-ID		online video social network.		user infor-	behavior, social relation
Qazinian et al. (2011)Identified tweets in which rumor is endorsed.Naïve Bayes Naïve BayesTweetsContent-based, network- based, Twitter specific memes.Chhabra et al. (2011)Using URLs static features, a method is developed to detect malicious websites.Naïve Bayes, Logistic Re- gression, DT, SVM-RBF, SVM-Linear, SVM-SigmoidMalicious URL dataset from Phish- tank'Grammar, Lexical, Vec- tors and Static.Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sionTweets and correspond- informationTopic engagement, Global engagement, Social repu- tation, Likability, Credi- bilityChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'BluffDiscourseRubin et al. (2015)A hybrid approach is pro- oped.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affect, Punctuation. newspapersAhmed et al. (2017) al cal cal at ricles.Developed n-gram based classi- fier to differentiate between fake ad real articles.Linguistic mod- elsNews postsTF-IDF, Doc2VecCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- k- brie deversemed subjects of suspi- cious news; satire, hoaxes, click- k- brie deversemed subjects of suspi- cious news; satire, hoaxes, click- k- brie deversemed, hoaxes, click- k- brie deversemed, hoaxes, click- k- brie deversemed, hoaxes, click- k- brie deversemed, hoaxes, click- k- bri	Qazinian et al. (2011)Identified tweets in which rumor is endorsed.Naïve BayesTweetssponse interactions.Chabar et al. (2011)Using URLs static features, a method is developed to detect malicious websites.Naïve Bayes, Logistic Re- gression, DT, SVM-RBF, SVM-Linear, SVM-SigmoidMalicious URL dataset from 'Phish- tank'Grammar, Lexical, V tors and Static.Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sionTweets and correspond- informationTopic engagement, Glc engagement, Social re tank'Chen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'BluffDiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative aff punctuation.Ahmed et al. (2015)Developed n-gram based classi- fier to differentiate between fak ad real articles.LinearSVMNews articles elsTF-IDF, Doc2Vec VectorizerCaetano et systemUsing textual data of articles, an systemSGD, PA, hultinomi- alXB, Gradient Bosting,News articles tweet vectorizerTF-IDF, Con Vectorizer				mation.	between users via video re-
Qazinian et al. (2011) Identified tweets in which rumor al. (2011) Naïve Bayes Tweets Content-based, network-based, network-based, network-based, metwork-based, metwork-based, network-based, network, network-based, network, networ	Qazinian al. (2011)Identified tweets in which rumor is endorsed.Naïve Bayes at (2011)TweetsContent-based, netwe based, Twitter spec memes.Chhabra al. (2011)Using URLs static features, a method is developed to detect malicious websites.Naïve Bayes, Logistic Re- gression, DT, SVM-RBF, SVM-RBF, SVM-RBF, SVM-SigmoidMalicious URL dataset from 'Phish- tank'Grammar, Lexical, V tors and Static.Gupta (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Reges- source sigmoidTweets and correspond- informationTopic engagement, Glc engagement, Social re tation, Likability, Cre bilityChen (2015)Analyzed coherence relations be tween deceptive and truthful network-based behavior data.VSMNews sam- ples from NPR's 'BluffRubin (2015)A hybrid approach is pro- posed combining linguistic and oped.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mational national <b< td=""><td></td><td></td><td></td><td></td><td>sponse interactions.</td></b<>					sponse interactions.
al. (2011) is endorsed. based, Twitter specific memes. Chhabra et al. (2011) Using URLs static features, a al. (2011) Naïve Bayes, memes. Malicious URL dataset from 'Phish-tank' Grammar, Lexical, Vec-tors and Static. al. (2011) method is developed to detect malicious websites. Logistic Regression, DT, SVM-RIBF, SVM-RIBF, SVM-Sigmoid Tweets and corresponding user information Topic engagement, Global engagement, Social reputation, Likability, Credibility (2013) Boston Marathon. Logistic Regression Tweets and corresponding user information Topic engagement, Social reputation, Likability, Credibility (2015) Analyzed coherence relations between deceptive and truthful news. VSM News samples from NPR's 'Bluff the Listener' Rubin et al. (2015) A hybrid approach is prooged. Linguistic, Network-based behavior data. SVM Sum and truthous and network-based behavior data. SVM Sum and truthous and trutho	al. (2011) is endorsed. based, Twitter spectiments. Chhabra et al. (2011) Using URLs static features, a method is developed to detect malicious websites. Naïve Bayes, Logistic Regression, DT, SVM-RBF, SVM-Innear, SVM-RBF, SVM-Sigmoid Malicious URL dataset from 'Phish-tank' tors and Static. Gupta et al. (2013) Analysis of Twitter content during Boston Marathon. Logistic Regression Tweets and corresponding user information Topic engagement, Social re billity Chen et al. (2015) Analyzed coherence relations be tween deceptive and truthful news. VSM News sam- ples from 'Phish-text's sentences Discourse Rubin et al. (2015) A hybrid approach is project combining linguistic and network-based behavior data. Linguistic, Net-work models Simple text sentences Bag of Words, n-gram mar, Negative affing antional network-based classi-fier to differentiate between fake ad real articles. (2015) Developed n-gram based classi-fier to differentiate between fake ad real articles. LinearSVM News articles TF-IDF, Doc2Vec afficient antional network-based classi-fier to gragmanda. (2017) fier to differentiate between fake ad real articles. SGD, PA, News articles TF-IDF, Corresponder and propaganda. Proposed system Using textual data of articles, an developed to detect fake arti-fier. SMM News articles TF-IDF, Corevec vectorizer, Hashi Vectorizer<	Qazinian et	Identified tweets in which rumor	Naïve Bayes	Tweets	Content-based, network-
Chabra et al. (2011)Using URLs static features, a developed to detect malicious websites.Naïve Bayes, Logistic Re- gression, DT, SVM-RBF, SVM-Linear, SVM-SigmidMalicious URL dataset from 'Phish- tank'Grammar, Lexical, Vec- tors and Static.Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sinTweets and correspond- ing user informationTopic engagement, Global engagement, Social repu- tation, Likability, Credi- bilityChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- oped.Linguistic, Net- work modelsUSMalicious under tal. A satire detection model is devel- oped.SVMUS and national newspapersAbsurdity, Humor, Gram- mar, Negative affect, Punctuation.Ahmed et al. (2017)Developed n-gram based classi- dareal articles.Linguistic mod- elsNews postsTF-IDF, Doc2VecCaetano et ki (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- ki ted mersener deLinguistic mod- elsNews postsTF-IDF, Doc2Vec	Chabra et al. (2011)understand method is developed to detect malicious websites.Naise News al. (2011)There is method is developed to detect malicious websites.Naise Re- gression, DT, SVM-RBF, SVM-Imear, SVM-SigmoidGrammar, Lexical, V tors and Static.Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sionTweets and correspond- informationTopic engagement, Glc engagement, Social re informationChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from work modelsDiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.SVMUS and antional network-based behavior data.SVMUS and canadian national newspapersBag of Words, n-gram mar, Negative affi Punctuation.Conroy et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews anticlesTF-IDFCaetano (2018)A predictive model was built to predict 4 subtypes of suspi- to sus wes; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews anticlesTF-IDF, Coc Vectorizer, Hashi Vectorizer, Hashi Vectorizer, Hashi Vectorizer	al. (2011)	is endorsed.			based, Twitter specific
Chhabra et al. (2011)Using URLs static features, a method is developed to detect malicious websites.Naïve Bayes, Logistic SVM-RBF, SVM-RBF, SVM-Linear, SVM-SigmoidMalicious URL dataset trom 'Phish- tank'Grammar, Lexical, Vec- tors and Static.Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sionTweets and correspond- ing user informationTopic engagement, Global correspond- ing user informationTopic engagement, Global correspond- ing user informationChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'BluffDiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSVMSimple text sentencesBag of Words, n-gram mar, Negative affect, Punctuation.Conroy et al. (2017)A satire detection model is devel- oped.SVMUS and A struct detection model is devel- ad real articles.LinearSVMNews articlesTF-IDFCaetano (2017)Eveloped n-gram based classi- fer to differentiate between fake ad real articles.Linguistic mod- elsNews postsTF-IDF, Doc2Vec	Chhabra et al. (2011)Using URLs static features, a method is developed to detect malicious websites.Naïve Logistic Re- gression, DT, SVM-RBF, SVM-SignoidMalicious URL dataset from 'Phish- tank'Grammar, Lexical, V tors and Static.Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.SVM-SignoidTweets and correspond- ing user informationTopic engagement, Glc engagement, Social re informationChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affe punctuation.Conroy et al. (2015)A satire detection model is devel- oped.SVMUS and canadian national network-based behavior data.SVMUS and canadian national network data.Conroy et al. (2017)Developed n-gram based classi- to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews articlesTF-IDF, Doc2VecCaetano to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.SGD, PA, Multinomi- alNB, Gradient Boosting,News articlesTF-IDF, Cor Vectorizer, Hashi Vectorizer					memes.
al. (2011) method is developed to detect malicious websites. Logistic server Logistic SVM-RBF, SVM-RBF, SVM-RBF, SVM-Sigmoid URL dataset from 'Phish- tank' tors and Static. Gupta et al. (2013) Analysis of Twitter content dur- ing Boston Marathon. Logistic Regres- sion Tweets and correspond- ing user Topic engagement, Global engagement, Social repu- tation, Likability, Credi- bility Chen et al. (2015) Analyzed coherence relations be- tween deceptive and truthful news. VSM News sam- ples from NPR's 'Bluff the Listener' Discourse Rubin et al. (2015) A hybrid approach is pro- posed combining linguistic and network-based behavior data. Linguistic, Net- work models Simple text sentences Bag of Words, n-gram mar, Negative affect, Punctuation. Conroy et al. (2015) A satire detection model is devel- oped. SVM US and canadian national network-based classi- dation. Absurdity, Humor, Gram- mar, Negative affect, Punctuation. Ahmed et al. (2017) Developed n-gram based classi- dat real articles. Linguistic mod- els News posts TF-IDF, Doc2Vec	al. (2011) method is developed to detect malicious websites. Logistic Re- gression, DT, SVM-RBF, SCM, SVM-RBF, SCD, SCD, SVM-RBF, SCD, SCD, SCD, SCD, SCD, SCD, SCD, SCD	Chhabra et	Using URLs static features, a	Naïve Bayes,	Malicious	Grammar, Lexical, Vec-
malicious websites.gression, DT, SVM-RBF, SVM-RBF, SVM-Linear, SVM-Sigmoidfrom 'Phish- tank'Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sionTweets and correspond- informationTopic engagement, Global engagement, Social repu- tation, Likability, Credi- bilityChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affect, Punctuation.Conroy et al. (2017)A satire detection model is devel- oped.SVMUS enadian national network-based behavior data.TF-IDFConroy et al. (2017)Developed n-gram based classi- dataLinearSVMNews articlesTF-IDF, Doc2VecCatano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- basit and merspandeLinguistic mod- elsNews postsTF-IDF, Doc2Vec	malicious websites.gression, DT, SVM-RBF, SVM-Linear, SVM-Sigmoidfrom 'Phish- tank'Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sionTweets and correspond- ing userTopic engagement, Glc engagement, Social re unformationChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'BluffDiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affe punctuation.Conroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian national news articlesAbsurdity, Humor, Gra mar, Negative affe punctuation.Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between faka ad real articles.LinearSVMNews articlesTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.SGD, PA, Multinomi- alNB, Gradient Boosting, Boosting,News articlesTF-IDF, Cor Vectorizer, Hashi Vectorizer, Hashi Vectorizer	al. (2011)	method is developed to detect	Logistic Re-	URL dataset	tors and Static.
SVM-RBF, SVM-Linear, SVM-Sigmoidtank'Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sionTweets and correspond- ing user informationTopic engagement, Global engagement, Social repu- tation, Likability, Credi- bilityChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affect, Punctuation.Conroy et al. (2015)A satire detection model is devel- oped.SVMUS and national newspapersAbsurdity, Humor, Gram- mar, Negative affect, Punctuation.Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.Linguistic mod- lesNews postsTF-IDF, Doc2VecCacatano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- basit and mersemende.Linguistic mod- elsNews postsTF-IDF, Doc2Vec	SVM-RBF, SVM-Linear, SVM-Sigmoidtank'Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sionTweets and correspond- ing user informationTopic engagement, Glc engagement, Social re tation, Likability, Cre bilityChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'Rubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesConroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian network-based behavior data.Absurdity, Humor, Gra mar, Negative affe national newspapersAhmed et al. (2017)Developed n-gram based classi- tier to differentiate between fake ad real articles.Linguistic mod- telsNews articlesTF-IDF, Doc2VecCataton et al (2018)A propaganda.SGD, PA, Multinomi- al MB, Gradient Bosting,News articlesTF-IDF, Cor Vectorizer, Hashi Vectorizer, Hashi Nultinomi-		malicious websites.	gression, DT,	from 'Phish-	
Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.SVM-Linear, SVM-SigmoidTweets and correspond- ing user informationTopic engagement, Global engagement, Social repu- ing user bilityChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text canadian national newspapersBag of Words, n-gram mar, Negative affect, Punctuation.Conroy et al. (2015)Developed n-gram based classi- ad real articles.SVMUS and canadian national newspapersAbsurdity, Humor, Gram- mar, Negative affect, Punctuation.Ahmed et al. (2017)Developed n-gram based classi- der al articles.LinearSVMNews postsTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- heit and newsparsendeLinguistic mod- elsNews postsTF-IDF, Doc2Vec	SVM-Linear, SVM-SigmoidSVM-Linear, SVM-SigmoidGupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sionTweets and correspond- ing user informationTopic engagement, Glc engagement, Social re bilityChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affe Punctuation.Conroy et al. (2015)Developed n-gram based classi- fier to differentiate between fale al real articles.LinearSVMNews articlesTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews articlesTF-IDF, Doc2VecProposed systemUsing textual data of articles, and is developed to detect fake arti- cles.SGD, PA, Multinomi- alNB, Gradient Boosting,News articlesTF-IDF, Cor			SVM-RBF,	tank'	
Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.SVM-SigmoidTweets and correspond- ing user informationTopic engagement, Global engagement, Social repu- tation, Likability, Credi- bilityChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affect, 	Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.SVM-SigmoidTweets and correspond- ing user informationTopic engagement, Glc engagement, Social re engagement, Social re information(2013)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSinple text sentencesBag of Words, n-gram mar, Negative afficient punctuation.Conroy et al. (2015)A satire detection model is devel- oped.SVMUS and national national newspapersAbsurdity, Humor, Gra mar, Negative afficient punctuation.Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.SGD, PA, Multinomi- alNB, Gradient BNR, Gradient alNB, Gradient BNS, Gradient BNS, GradientTF-IDF, Cou Vectorizer, Hashi Vectorizer, Hashi Vectorizer			SVM-Linear,		
Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sionTweets and correspond- informationTopic engagement, Global engagement, Social repu- tation, Likability, Credi- bilityChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from work modelsDiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affect, punctuation.Conroy et al. (2015)A satire detection model is devel- oped.SVMUS and canadian newspapersAbsurdity, Humor, Gram- mar, Negative affect, punctuation.Ahmed et al. (2017)Developed n-gram based classi- fer to differentiate between fake ad real articles.Linguistic mod- news papersNews postsTF-IDF, Doc2VecCaetano et (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bicit and enversemende.Linguistic mod- elsNews postsTF-IDF, Doc2Vec	Gupta et al. (2013)Analysis of Twitter content dur- ing Boston Marathon.Logistic Regres- sionTweets and correspond- informationTopic engagement, Gle engagement, Social re information(2013)Ing Boston Marathon.SionSionCorrespond- informationengagement, Social re tation, Likability, Cre bilityChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and oped.Linguistic, Net- work modelsSimple text canadian national network-based behavior data.Bag of Words, n-gram mar, Negative affi Punctuation.Conroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian national national network-based behavior data.Theiras and Punctuation.Catatano et al. (2018)Developed n-gram based classi- d real articles.LinearSVMNews articlesTF-IDF, Doc2VecCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.SGD, PA, Multinomi- alNB, Gradient Bosting,News articlesTF-IDF, Cour Vectorizer, Hashi Vectorizer, Hashi Vectorizer			SVM-Sigmoid		
(2013)ing Boston Marathon.sioncorrespond- ing user informationengagement, Social repu- tation, Likability, Credi- bilityChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affect, Punctuation.Conroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian national network-based behavior data.Absurdity, Humor, Gram- mar, Negative affect, Punctuation.Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.Linguistic mod- linguistic mod- elsNews postsTF-IDF, Doc2VecCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- brit and newspapersLinguistic mod- elsNews postsTF-IDF, Doc2Vec	(2013)ing Boston Marathon.sioncorrespond- informationengagement, Social re tation, Likability, Cre bilityChen et al.Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al.A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative aff Punctuation.Conroy et al.A satire detection model is devel- oped.SVMUS and Canadian national network-based behavior data.SVMUS and Canadian national network-based behavior data.Conroy et al.Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCactano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews articlesTF-IDF, Doc2VecProposed using textual data of articles, an systemUsing textual data of articles, an is developed to detect fake arti- cles.SGD, PA, Multinomi- alNB, Gradient Boosting,News articlesTF-IDF, Cour	Gupta <i>et al.</i>	Analysis of Twitter content dur-	Logistic Regres-	Tweets and	Topic engagement, Global
Image: Chen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affect, Punctuation.Conroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian national newspapersAbsurdity, Humor, Gram- mar, Negative affect, Punctuation.Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.Linguistic mod- elsNews postsTF-IDF, Doc2VecCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- brit and newspapersLinguistic mod- elsNews postsTF-IDF, Doc2Vec	Image: Chen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'BluffDiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram sentencesConroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian national newspapersAbsurdity, Humor, Gra mar, Negative affe Punctuation.Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.Linguistic mod- elsNews postsTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews articlesTF-IDF, Doc2Vec vectorizer, Hashi vectorizer, Hashi vectorizer, Hashi vectorizer	(2013)	ing Boston Marathon.	sion	correspond-	engagement, Social repu-
Image: Chen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affect, national newspapersConroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian newspapersAbsurdity, Humor, Gram- mar, Negative affect, national newspapersAhmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.Linguistic mod- elsNews postsTF-IDF, Doc2VecCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- heit and neuronemed.Linguistic mod- elsNews postsTF-IDF, Doc2Vec	Image: constraint of the second systeminformationbilityChen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative aff punctuation.Conroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian national network-based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews articlesTF-IDF, Doc2VecProposed systemUsing textual data of articles, an s developed to detect fake arti- cles.SGD, PA, Multinomi- alNB, Gradient Boosting,News articlesTF-IDF, Courser, Hashi Vectorizer				ing user	tation, Likability, Credi-
Chen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affect, national network-based behavior data.Conroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian network-based classi- fier to differentiate between fake ad real articles.Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews postsTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bei and neuronerandeLinguistic mod- elsNews postsTF-IDF, Doc2Vec	Chen et al. (2015)Analyzed coherence relations be- tween deceptive and truthful news.VSMNews sam- ples from NPR's 'Bluff the Listener'DiscourseRubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, Negative affe punctuation.Conroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian national network-based behavior data.Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano et al. (2018)A predicitve model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.SGD, PA, Multinomi- alNB, Gradient BNG, Gradient BNG, Gradient BNG, GradientNews articlesTF-IDF, Courser				information	bility
(2015)tween deceptive and truthful news.ples ples from NPR's 'Bluff the Listener'Rubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram sentencesConroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian newspapersAbsurdity, Humor, Gram- mar, Negative affect, Punctuation. newspapersAhmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- heit and neurogeneticeLinguistic mod- elsNews postsTF-IDF, Doc2Vec	(2015)tween deceptive and truthful news.ples ples from NPR's 'Bluff the Listener'Rubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram sentencesConroy et al. (2015)A satire detection model is devel- oped.SVMUS canadian national newspapersAbsurdity, Humor, Gra mar, Negative affi Punctuation.Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews articlesTF-IDF, Doc2VecProposed systemUsing textual data of articles, an sideveloped to detect fake arti- cles.SGD, PA, Multinomi- alNB, Gradient Boosting,News articlesTF-IDF, Cour Vectorizer, Hashi Vectorizer	Chen <i>et al.</i>	Analyzed coherence relations be-	VSM	News sam-	Discourse
news.NPR's 'Bluff the Listener'Rubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram sentencesConroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian national network-based behavior.Absurdity, Humor, Gram- mar, Negative affect, national network-based behavior.Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.Linguistic mod- lesNews articlesCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- beit and prome reacheLinguistic mod- elsNews postsTF-IDF, Doc2Vec	news.NPR's 'Bluff the Listener'Rubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram mar, negramConroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian national network-based classi- fier to differentiate between fake ad real articles.SVMUS and Canadian national newspapersAbsurdity, Humor, Gra mar, Negative affi Punctuation.Caetano al. (2018)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews articlesTF-IDF, Doc2VecProposed systemUsing textual data of articles, an setuel data of articles, an cles.SGD, PA, Multinomi- alNB, Gradient Boosting,News articlesTF-IDF, Cour	(2015)	tween deceptive and truthful		ples from	
Rubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram Bag of Words, n-gram sentencesConroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian national network-based behavior.Absurdity, Humor, Gram- mar, Negative affect, national newspapersAhmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and menoramed.Linguistic mod- elsNews postsTF-IDF, Doc2Vec	Rubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gram Bag of Words, n-gram sentencesConroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian national network-based classi- fier to differentiate between fake ad real articles.SVMUS and Canadian national newspapersAbsurdity, Humor, Gra mar, Negative affe Punctuation.Caetano et al. (2018)Developed n-gram based classi- fier to differentiate between fake ad real articles.Linguistic mod- elsNews articlesTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews articlesTF-IDF, Doc2VecProposed systemUsing textual data of articles, an is developed to detect fake arti- cles.SGD, PA, Multinomi- alNB, Gradient Boosting,TF-IDF, Courser, Vectorizer		news.		NPR's 'Bluff	
Rubin et al. (2015)A hybrid approach is pro- posed combining linguistic and network-based behavior data.Linguistic, Net- work modelsSimple text sentencesBag of Words, n-gramConroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian national newspapersAbsurdity, Humor, Gram- mar, Negative affect, Punctuation.Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and prenegrandaLinguistic mod- elsNews postsTF-IDF, Doc2Vec	Rubin et al.A hybrid approach is proposed combining linguistic and network-based behavior data.Linguistic, Network modelsSimple text sentencesBag of Words, n-gram(2015)posed combining linguistic and network-based behavior data.A satire detection model is developed.SVMUS and Canadian national punctuation.Absurdity, Humor, Grama, Negative afferences(2015)oped.LinearSVMNews articlesTF-IDF(2017)fier to differentiate between fake ad real articles.Linguistic modelsNews postsTF-IDF, Doc2VecCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspicious news; satire, hoaxes, clickbait and propaganda.Linguistic modelsNews articlesTF-IDF, Doc2VecProposedUsing textual data of articles, an systemSGD, PA, News articlesTF-IDF, Courser, Hashi ultinomial alNB, Gradient Boosting,TF-IDF, Courser				the Listener'	
(2015)posed combining linguistic and network-based behavior data.work modelssentencesConroy et al.A satire detection model is devel- oped.SVMUS and Canadian national newspapersAbsurdity, Humor, Gram- mar, Negative affect, national newspapersAhmed et al.Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesCaetano al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- hait and menomenandeLinguistic mod- elsNews postsTF-IDF, Doc2Vec	(2015)posed combining linguistic and network-based behavior data.work modelssentencesConroy et al. (2015)A satire detection model is devel- oped.SVMUSand Canadian national newspapersAbsurdity, Humor, Gra mar, Negative affi Punctuation.Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews articlesTF-IDF, Doc2VecProposed systemUsing textual data of articles, an sig developed to detect fake arti- cles.SGD, PA, Multinomi- alNB, Gradient Boosting,News articlesTF-IDF, Cour Vectorizer, Hashi Vectorizer	Rubin <i>et al.</i>	A hybrid approach is pro-	Linguistic, Net-	Simple text	Bag of Words, n-gram
network-based behavior data.Image: constraint of the section model is developed oped.SVMUS and Canadian mar, Negative affect, national newspapers(2015)oped.Image: constraint oped.News articlesPunctuation.Ahmed et al.Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- heit and pregram dateLinguistic mod- elsNews postsTF-IDF, Doc2Vec	network-based behavior data.News articlesConroy et al. (2015)A satire detection model is devel- oped.SVMUS and Canadian national newspapersAbsurdity, Humor, Gra mar, Negative affi punctuation. newspapersAhmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews articlesProposed systemUsing textual data of articles, an is developed to detect fake arti- cles.SGD, PA, Multinomi- alNB, Gradient Boosting,News articlesTF-IDF, Cour Vectorizer	(2015)	posed combining linguistic and	work models	sentences	
Conroy et al.A satire detection model is devel- oped.SVMUS Canadian national newspapersAbsurdity, Humor, Gram- mar, Negative affect, Punctuation.(2015)oped.Image: Canadian newspapersmar, Negative affect, punctuation.Ahmed et al.Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- heit and preparendeLinguistic mod- elsTF-IDF, Doc2Vec	Conroy et al. (2015)A satire detection model is devel- oped.SVMUSand Canadian national newspapersAbsurdity, Humor, Gra mar, Negative aff Punctuation.Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews postsTF-IDF, Doc2VecProposedUsing textual data of articles, an systemSGD, PA, efficient multi-level voting model is developed to detect fake arti- cles.SGD, PA, Multinomi- alNB, Gradient Boosting,TF-IDF, Cour Vectorizer		network-based behavior data.			
(2015)oped.Canadian national newspapersmar, Negative affect, Punctuation.Ahmed et al.Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- heit and prenegrandeLinguistic mod- elsNews postsTF-IDF, Doc2Vec	(2015)oped.Canadian national newspapersmar, Punctuation.Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews postsTF-IDF, Doc2VecProposed systemUsing textual data of articles, an si developed to detect fake arti- cles.SGD, PA, Multinomi- alNB, Gradient Boosting,News articlesTF-IDF, Cour Vectorizer	Conroy <i>et al.</i>	A satire detection model is devel-	SVM	US and	Absurdity, Humor, Gram-
Ahmed et al.Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano et al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- heit and pregram deLinguistic mod- elsNews postsTF-IDF, Doc2Vec	Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews postsTF-IDF, Doc2VecProposed systemUsing textual data of articles, an si developed to detect fake arti- cles.SGD, PA, Multinomi- alNB, Gradient Boosting,News articlesTF-IDF, Court	(2015)	oped.		Canadian	mar, Negative affect,
Ahmed et al.Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetanoetA predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- heit and pregrammedeLinguistic mod- elsNews postsTF-IDF, Doc2Vec	Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano al. (2018)A predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews postsTF-IDF, Doc2VecProposedUsing textual data of articles, an is developed to detect fake arti- cles.SGD, PA, alNB, Gradient Boosting,News articlesTF-IDF, Cou Vectorizer				national	Punctuation.
Ahmed et al. Developed n-gram based classi- LinearSVM News articles TF-IDF (2017) fier to differentiate between fake ad real articles. al real articles. TF-IDF, Doc2Vec TF-IDF, Doc2Vec Caetano et A predictive model was built to predict 4 subtypes of suspicious news; satire, hoaxes, click- Linguistic mod- News posts TF-IDF, Doc2Vec	Ahmed et al. (2017)Developed n-gram based classi- fier to differentiate between fake ad real articles.LinearSVMNews articlesTF-IDFCaetano al. (2018)et to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews postsTF-IDF, Doc2VecProposed systemUsing textual data of articles, an is developed to detect fake arti- cles.SGD, PA, Multinomi- alNB, Gradient Boosting,News articlesTF-IDF, Cou Vectorizer				newspapers	
(2017) fier to differentiate between fake ad real articles. Image: Caetano et al. (2018) A predictive model was built Linguistic mod- els News posts TF-IDF, Doc2Vec al. (2018) to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- hoit and preparende els Image: Caetano et els Image: Caetano et els Image: Caetano et els	(2017)fier to differentiate between fake ad real articles.Linguistic mod- elsTF-IDF, Doc2VecCaetanoetA predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews postsTF-IDF, Doc2VecProposedUsing textual data of articles, an systemSGD, PA, Multinomi- alNB, Gradient Boosting,News articlesTF-IDF, Court	Ahmed <i>et al.</i>	Developed n-gram based classi-	LinearSVM	News articles	TF-IDF
ad real articles. ad real articles. Tr-IDF, Doc2Vec Caetano et A predictive model was built Linguistic mod- News posts TF-IDF, Doc2Vec al. (2018) to predict 4 subtypes of suspi- els Els Els Els bait and preparende Els Els Els Els Els	ad real articles.Image: Caetano et al. (2018)ad real articles.Image: Caetano et al. (2018)A predictive model was built to predict 4 subtypes of suspicious news; satire, hoaxes, click-bait and propaganda.Linguistic model elsNews postsTF-IDF, Doc2VecProposedUsing textual data of articles, an systemSGD, PA, efficient multi-level voting model is developed to detect fake articles.SGD, PA, efficient galaxiesNews articlesTF-IDF, Courser, HashiesVectorizerBoosting,Boosting,News articlesNews articlesNews articles	(2017)	fier to differentiate between fake			
Caetano et A predictive model was built Linguistic mod- News posts TF-IDF, Doc2Vec al. (2018) to predict 4 subtypes of suspi- els els TF-IDF, Doc2Vec	CaetanoetA predictive model was built to predict 4 subtypes of suspi- cious news; satire, hoaxes, click- bait and propaganda.Linguistic mod- elsNews postsTF-IDF, Doc2VecProposedUsing textual data of articles, an systemSGD, PA, Multinomi- alNB, Gradient Boosting,News articlesTF-IDF, Vectorizer		ad real articles.			
al. (2018) to predict 4 subtypes of suspi- cious news; satire, hoaxes, click-	al. (2018) to predict 4 subtypes of suspicious news; satire, hoaxes, clickbait and propaganda. els els els Proposed Using textual data of articles, an efficient multi-level voting model is developed to detect fake articles. SGD, PA, News articles TF-IDF, Court Vectorizer, Hashing Multinomi- Vectorizer, Hashing is developed to detect fake articles. Boosting, Vectorizer	Caetano et	A predictive model was built	Linguistic mod-	News posts	TF-IDF, Doc2Vec
cious news; satire, hoaxes, click-	cious news; satire, hoaxes, click- bait and propaganda.cious news; satire, hoaxes, click- bait and propaganda.ProposedUsing textual data of articles, an efficient multi-level voting model is developed to detect fake arti- 	al. (2018)	to predict 4 subtypes of suspi-	els	-	
heit and menomenda	bait and propaganda.SGD, PA,News articlesTF-IDF,CourseProposedUsing textual data of articles, an efficient multi-level voting modelSGD, PA,News articlesTF-IDF,Coursesystemefficient multi-level voting modelMultinomi- alNB, GradientVectorizer, VectorizerHashi Vectorizercles.Boosting,VectorizerVectorizer		cious news; satire, hoaxes, click-			
balt and propaganda.	Proposed systemUsing textual data of articles, an efficient multi-level voting model is developed to detect fake arti- cles.SGD, Multinomi- alNB, Gradient Boosting,PA, News articlesTF-IDF, Vectorizer, Hashi Vectorizer		bait and propaganda.			
Proposed Using textual data of articles, an SGD, PA, News articles TF-IDF, Count-	system efficient multi-level voting model Multinomi- is developed to detect fake arti- cles. Boosting, Vectorizer, Hashi	Proposed	Using textual data of articles, an	SGD, PA,	News articles	TF-IDF, Count-
system efficient multi-level voting model Multinomi- Vectorizer, Hashing-	is developed to detect fake arti- cles. Boosting, Vectorizer	system	efficient multi-level voting model	Multinomi-		Vectorizer, Hashing-
is developed to detect fake arti- alNB, Gradient Vectorizer	cles. Boosting,	-	is developed to detect fake arti-	alNB, Gradient		Vectorizer
cles. Boosting,	0,		cles.	Boosting,		
	DT,AdaBoost			DT,AdaBoost		

Table 1: Comparative analysis of research studies



Fig. 1: Architecture of the proposed automatic fake news detection system

Overall statistics of our three collected datasets are discussed in Table 2. To analyze the length distribution of the titles for fake and real articles, Figure 2 is visualized. Where the X-axis labeled as 'title' represents the number of terms used in news article title's/headlines whereas Y-axis represents the corresponding number of articles having the same length distribution. A conclusion can be drawn from the mean distribution by visualising Figure 2 that the length of the title's or headlines of fake news articles is often longer than real news articles.

To further analyze the length distribution of the body content in articles, Figure 3 is observed, where the X-axis labeled as 'text' represents the number of terms used in news article texts whereas Y-axis represents the corresponding number of articles having the same length distribution.

It has been analyzed that the length of body/text of real news articles are often longer than fake news articles as shown in Figure 3(a) and 3(b) but for Reuters corpus the length of real news articles is more than fake news articles as seen in Figure 3(c).

The statistics of the mean distribution for Figure 2 and Figure 3 are compared in Table 3. In general, a conclusion can be drawn after analysing different datasets that the headlines of fake articles are longer in length and have shorter body content than real articles published on social networking [42] sites.

4.2 Pre-processing

In the pre-processing phase, non-semantic words such as prepositions, conjunctions, pronouns, *etc.*, also known as stop words are removed from the textual document as they provide very little or no information about fake content of an article. The redundant data in form of textual strings is removed from the document using a regular expression *(regex)* in the next step as shown in Figure 4. The *regex* and *pandas* library has been used to perform the pre-processing task [43]. *Regex* library has been used in python to define a search pattern using a sequence of characters whereas *dropna* method from *pandas* is used for cleaning the missing values in python DataFrame. We have used *random_state* function to select the entries from the dataset which is further used to split training and testing data points as it is used to split the data randomly.

To avoid overfitting, three standard splits (70:30, 67:33, 60:40) were used to perform the experiment. When the first standard split (70:30) was performed, it was observed that the data point dealt with an issue of underfitting. During the second split (60:40), overfitting of data was analyzed. Whereas, the third split (67:33) gave the best predicted line covering the majority of data points in the graph, so a standard split of 67:33 was chosen. The training data is then fed to feature generation phase as discussed in the next section.

4.3 Feature Generation

In order to extract numerical features from a textual document, tokenization, counting and normalization is done. During tokenization, each word is given a unique integer ID, following which occurrence of tokens is counted and then normalization of such tokens takes place. The whole process of converting the textual document into numerical feature vector is called as vectorization. Together this strategy (tokenization, counting, normalization) is called as 'Bag of n-grams' or 'Bag of Words' where n represents the continuous sequence of terms [44]. The three feature extraction techniques as shown in Figure 5 for retrieving features from the textual content of an article are Count-Vectorizer, TF-IDF and Hashing-Vectorizer which use CountVectorizer, TfidfVectorizer and Hashing Vectorizer classes from feature_extraction library of python, respectively.

4.3.1 Count Vectorizer (CV)

Count vectorizer is absolutely based on count of the word occurrences in the document. In count vectorizer technique, both counting the occurrences of tokens and

0	Corpus	Total	article	Clean	ed art	icles	Real	art	\mathbf{ticles}	Fake	articles	Pub	lishe	ed y	ear
Nev	vs Trends	77	95		6335			3171	L		3164		201	7	
]	Kaggle	40	48		3983			1865)		2118		201	.7	
F	Reuters	215	578		19969			9622	2	1	0347		200)4	
0.025	Len	gth Distribution	REA			Length	Distribution	ı	REAL	ſ	L	ength Distril	bution		DEAL
0.020 -			FAK	E 0.035 - 0.030 - 0.025 -					FAKE	0.030 - 0.025 - 0.020 -	h				FAKE
0.015 -				0.020 - 0.015 - 0.010 -						0.015 - 0.010 -					
0.005	0 50 100	150 200	250 30	0.005	0 100	200 3	400	500	600 700	0.005 -	0 50 1	00 150	200	250	300

Table 2: Statistics of collected corpora

ttle Headline ttle

Fig. 2: Length distribution of headline for fake and real articles on (a) News Trends (b) Kaggle (c) Reuters corpora



Fig. 3: Length distribution of text for fake and real articles on (a) News Trends (b) Kaggle (c) Reuters corpora



Fig. 4: Steps performed during the pre-processing phase

tokenization process is performed. Count vectorizer has many parameters for refining the type of features. One can build features using any of the three parameters, unigram $(min_df=1)$, bigram $(min_df=2)$ and trigrams $(min_df=3)$. We have used min_df as 1 for our experiment. Here, each vector (term) in a document represents the individual feature name and its occurrence is depicted through a matrix to make it easier to understand as shown in Table 4. It has been observed from the above table, after pre-processing phase, the terms retrieved from the documents are represented as vectors on top of the sparse matrix and the frequency of terms in particular document is represented through count occurrences. Tag clouds of top 30 features retrieved after executing CV method is shown in Figure 6. It was observed that words like corruption, attacking, islamic, obama, losing, com are seen under fake tag clouds.

CV also counts the number of words occurring more frequently in the document, which may overshadow the words which occur less frequently but may have more importance to the document feature. This limitation of

Table	e 3: Mean distribution of	f labeled articles for New	rs Trends, Kaggle and R	euters corpora
Corpus	Mean distribution	Mean distribution	Mean distribution	Mean distribution
	of title labeled as	of title labeled as	of text labeled as	of text labeled as
	fake	real	fake	real
News Trends	69.18	61.38	4121.04	5292.16
Kaggle	62.47	57.32	2380.82	3544.84
Reuters	81.79	71.29	5156.08	4540.26



Fig. 5: Conversion of textual content into numerical vectors through TF-IDF, Count-vectorizer and Hash-vectorizer feature extraction techniques

Doc	ument	Narendra	elections	vote	politics	Punjab	BJP	candidate
D	loc1	0	1	2	0	0	2	1
D	loc2	4	0	1	0	0	1	0
D	loc3	1	3	2	0	1	0	1

Table 4: Sparse matrix representation using a Count-Vectorizer feature extraction technique

CV can be handled by using TF-IDF feature extraction technique as explained below.

4.3.2 Term frequency- Inverse Document frequency (Tf-Idf)

Tf-Idf is a weighing matrix which is used to measure the importance of a term (count + weight) to a document in a dataset. The tokens retrieved from the textual data using both Tf-Idf and CV techniques are same, but weights assigned to the tokens of both techniques are different. Tf-Idf is composed of two metrics, named as, term frequency (*tf*) and inverse document frequency (*idf*). Tf-Idf is represented by Eq. (1).

$$tfidf = tf(t,d) \times idf(t,d) \tag{1}$$

Here, Term Frequency denoted as tf and is calculated from the count (c), term (t) in document (d) and represented as $tf(t,d) = c_{td}$. The frequency of occurrence of words to a binary feature is converted by using 1 (present in document) and θ (not present in document). The frequencies can be normalized using average and logarithms. The inverse document frequency (idf) for a word w in document text (t) as computed by Eq. (2).

$$idf(t,d) = 1 + \log \frac{T}{(1+df(t))}$$
 (2)

Here, T represents the total document count in our corpus and df(t) represents the count of the number of documents where the term t is present. The product of two measures will help to compute *tfidf*. Euclidean's normalized form is used to calculate the final Tf-Idf

nomination continue 56. corruption democrat fox demo attacks deal debate october october posts print COM losing comments source hillary article step gop friday cabinet let witter jobs Sen transition held trade article^{war} podesta nov snip election podesta nov photo ayotte oct wars mosul deal debate march^{tuesday} . conservatives 11jewish^{step} entire ⁰⁸28share^{watch} rushparkerisrael demolish ditch fits takes gaining ohio fits takes continue march momentum cruz islamic jobs reuters email print don recent 2016 12 arrivals ²⁰¹⁶ hillary paris said said sunday hillary corporate recovery november fbi conference (a) (b) (a) (b) talked president reuters^{Survey} said family donald cnn rex^{world} bbc late unfolds shooting cookies Secretary harvey emigrate http articles tell coming https woman world hollywood harvey washington consumers contempt sorry great warning film pres americans publish cookies bbC^{family} said^{debate} Secretary rex authorities http_{thing} army contempt attacking warning articles fact learn legalization dan bothers www commences danger players unfolds. film_____says_night_woman trump www bothers Sorry war scandal danger learnworse chat shooting talked survey authorities chat players army destructive https emigrate worse scandal consumers m destructive co wednesday happening wednesday messenger com content (d) (d) (c) (c) inauguration support guest november administration did saturday twitter pamkeynen mr march percent schwarzenegger dominic mcgswipe rock dakota recent election pixabay Source comments hillary ^{ap} thereâ obamaâ sel k 2 slamic percent twitt reported drmm president leaders won sunday Sa 2017 islamic le administration lost october al massive oct 016documentary arrivals and continues documentary arrivals and continues novembershare 2016 turbeville brandon watch are ^{II ms} œi follow ^{IS} WON sunday said sive oct longer sunday friday print hillary ^{non} source com ^{old} don. administrationâ twitter administrationâ don . ewe march october illuminati york follow continues (e) (f) (e) (f)

Fig. 6: Top 30 (a) News Trends fake, (b) News Trends real, (c) Kaggle fake, (d) Kaggle real, (e) Reuters fake, (f) Reuters real word clouds using count-vectorizer feature extraction technique

metric as given by Eq. (3).

$$tfidf = \frac{tfidf}{||tfidf||} \tag{3}$$

Here ||tfidf|| is the Euclidean norm. Tag clouds of top 30 features for fake and real articles using TF-IDF feature extracting technique is shown in Figure 7. It was observed that words like *www*, *danger*, *sorry*, *wars*, *islamic* are most common under fake news articles. The difference between count-vectorizer and TF-IDF approach is that the tokens retrieved from textual data are same but both have different weights assigned to each token being extracted.

Fig. 7: Top 30 (a) News Trends fake, (b) News Trends real, (c) Kaggle fake, (d) Kaggle real, (e) Reuters fake, (f) Reuters real word clouds using TF-IDF feature extraction technique

4.3.3 Hashing Vectorizer (HV)

It is a memory-efficient technique. Unlike previous two techniques, tokens are stored as a string and here hashing trick is applied to encode the features as numerical indexes. Let us discuss the concept of HV using an example shown in Figure 8. When data is entered, the hashed attributes of data are retrieved. The hashed terms like Trump, election and politics are extracted from the document. In the next step, the hashing trick is applied on the hashed attributes, where a *Murmurhash3* function is applied on the hashed terms to generate a random number. Further, the assigned random numbers are divided by 8 and are stored in different keys such as k2, k3, k4 based on the remainders retrieved after applying the *murmurhash3* function which is used for hash-based lookup. There is a possibility of collision

Documents	Murmurhash3	Divide by	Remainder		Keys	Values
Triump	3452896710	8	3		k 0	-
election	3125690034	8	2	7	k1	-
politics	3567100209	8	4	1	k2	election
				./*	k3	Triump
				1	k4	politics

Fig. 8: Hashing trick on textual data using Murmurhash3 function to get values in a specific range

when data have equal hashed attributes.

Let us suppose in our example document, we have Trump and politics words as an important keys which are seen more than once, thus causing collisions at k3, k4 positions. The collided values are then occupied by other vacant positions in a set of documents. Such collision processing is dealt with parallel processing. The process is conceptually explained in Figure 9. These six



Fig. 9: Entry of redundant data into a hash table during parallel processing

terms are assigned six keys as shown in Figure 9 and are entered in the hash table. The hash values of keys k1, k3, k6 are same, *i.e.* Trump; k4 and k5 are same *i.e.* politics, but rest have different values. Due to collision k1, k3 and k6 cannot be placed in same set (S1). To enable parallel processing, k1, k3 and k6 are placed in different sets. No two same hash values can be placed in a single set. Different keys like k1, k2 and k4 can be placed in the same set (S1) as these keys have different values. Values at keys, k3 and k6 are same so cannot be processed parallelly, therefore are processed in different sets. The values in S1, S2, S3 are organized into vectors (numerical features) and can be processed using vector operations.



Fig. 10: Classification phase

The drawback of HV is that there is no way to get the feature names from feature indices, *i.e.*, the inverse transform cannot be used to compute the most important feature names through Hash-Vectorizer, unlike rest two methods.

5 Classification Algorithms

The processed dataset retrieved after pre-processing and feature extraction phase is then fed to the classification phase for the identification of fake news article. In this paper, six machine learning techniques, *i.e.*, Naïve Bayes (MultinomialNB), Support Vector Machine (Support Vector Classifier (SVC), NuSVC, LinearSVC), Decision Tree (CART), Linear (Passive Aggressive (PA), Stochastic Gradient Descent (SGD), Logistic Regression (LR)), Neural Network (Multi-Layer Perceptron (MLP)) and Ensemble models (AdaBoost, Gradient Boosting, Voting) have been applied as shown in Figure 10.

Classifier functions help to map the input feature vectors $f \in F$ to output labels $l \in \{1, 2, 3, \ldots, n\}$, where F is the feature space. The feature space is represented as, $F = \{Fake, Real\}^R$, where R is the real number. Our aim is to learn the classifier function from a labeled training data.

5.1 Naïve Bayes (NB)

It is a type of probability classifier. It works on Bayes Theorem and handles both categorical and continuous variables. NB assumes that every pair of features with labeled value is independent of each other. Given a collection of D documents from news articles, $D_i = \{d_1, d_2, ..., d_n\}$, where each document consists of T terms such as $D_i = \{t_1, t_2, ..., t_m\}$. Then the probability of D_i occurrence in class label C_l is given by Eq. (4).

$$P(C_{l}|D_{i}) = P(C_{l}) \prod_{n=1}^{m} P(d_{n}|C_{l})$$
(4)

Here, the conditional probability of term t_m present in a document of class label C_l and the prior probability of document occurring in class label C_l is denoted by $P(C_l)$.

Multinomial Naïve Bayes (MutinomialNB) is a type of Naïve Bayes algorithm used for text classification. Data used in text classification to apply MultinomialNB can be represented as *TF-IDF* vectors, hashing vectors and count vectors. The feature vectors $V_f = (V_{f1}, V_{f2}, ..., V_{fn})$ are parameterized for each class C_n in the distribution, where *n* represents the feature numbers. The likelihood of observing V_f is given by Eq. (5).

$$P(C_n|V_f) = \frac{X_i!}{\prod_{n=1}^p f_n!} \prod_{n=1}^P V_n f_n$$
(5)

Here, f_n is the number of times the n^{th} feature has occurred in the document, X_i is the number of draws taken from the bag of features. $V_n^{f_n}$ and $f_n!$ are computed from training data.

5.2 Linear Model

The linear model helps to classify the group by making linear combinations of feature vectors. Linear classifiers work well with many features but it works best for document (features are extracted from the text) classification. If \mathbf{v} is the input feature vector to the classifier, then the resultant score is given by Eq. (6).

$$s = f(\mathbf{vw}) = f(\sum_{i} w_i v_i) \tag{6}$$

here \mathbf{w} is the weight of a feature vector and the function f gives the desires output of two vectors. The three linear models used in this paper are Passive Aggressive (PA), Stochastic Gradient Descent (SGD) and Logistic Regression (LR) classifiers. The PA algorithm has similar behavior with perceptron classifier in terms of learning rate but has dissimilar behaviour in terms of regularization parameter. The PA classifier is equivalent to PA-1 [34] when the loss parameter is *hinge* and PA-ll [34] when the loss parameter *squared_hinge*. Second linear model used in this paper is SGD. The SGD model is updated with the decreasing learning rate after each sample interval and by default the loss parameter used in this paper is *hinge*. The classifier also allows minibatch learning. The LR model can be used for both multi and binary problem classification. Any other input format apart from *float64* gets converted in this classifier. All three linear models discussed above can take both sparse and dense matrix as their input.

5.3 Support Vector Machine (SVM)

SVM works on Structural Risk Minimization Principle. It is defined by a 'best' separating hyperplane and is also called as a discriminative classifier. Through the SVM model, feature vectors retrieved from text document of news articles are represented as points in feature space. Then the feature vectors are mapped in such a way that a wide gap is visible to perform linear classification. In our dataset the feature vectors are marked by making two categories, $C = \{Fake, Real\}$, then the training classifier builds a model which assigns new feature vectors to both defined categories.

SVM classes such as Linear Support Vector Classifier (LinearSVC), Nu-Support Vector Classifier (NuSVC), Support Vector Classifier (SVC) are used for performing classification on the dataset. NuSVC and SVC are almost similar but use slightly different sets of parameters and their mathematical formulations also vary. Whereas LinearSVC is another type of Support Vector Classification (SVC) and uses the case of linear kernel. All three classes take input in the form of two arrays with array X having 2-dimensional size [sample_number, feature_vectors] to handle training data and array Y having 2-dimensional size [category_label, sample_number]. The Decision function is the same for both SVC and NuSVC is given by Eq. (7).

$$sgn(\sum_{f=1}^{n} y_f \alpha_f K(V_f, V) + \mu)$$
(7)

where V_f are the training feature vectors, f = 1, 2, ..., nin two categories. $K(V_f, V)$ is the kernel and $y_f \alpha_f$ is the dual coefficient parameter which holds support vectors and an independent intercept term μ . The only difference between SVC $(C=[0,\infty])$ and NuSVC (C=[0,1]) is seen from parameter C which is the penalty parameter of the error term. The class LinearSVC supports both sparse and dense input and is implemented in terms of liblinear so is more flexible in terms of the loss function and penalties, and scales better to large testing samples.

5.4 Neural Network (NN)

Neural networks are composed of highly interconnected neurons to solve specific problem parallelly. In this paper, Multi-Layer Perceptron (MLP) is implemented on our collected dataset. The classifier can be trained on either regression or classification dataset. The feature vectors $V_f = \{v_1, v_2, \ldots, v_n\}$ are retrieved after feature extraction phase and through training dataset the classifier learns a given function in Eq. (8).

$$f(n): Ri \to Ro \tag{8}$$

where *i* is the dimensions for input and *o* is the dimensions for an output. In MLP there can be one or more non-linear layer (hidden layer) between the input and output layer. The input layer made up of neurons, where each neuron represents the input feature which is fed into the hidden layer. Then the hidden layer computes the weighted summation $w_1v_1 + w_2v_2 + \ldots + w_iv_i$, followed by function f(n). The output value is given by the last hidden layer and is received by the output layer.

5.5 Decision Tree (DT)

The DT classifiers can be used for both regression and classification. The classifier predicts the target variable by learning the feature data and dividing the area into sub-regions. Based on two criteria multiple features are divided, one is a measure of impurity and other is information gain. In our dataset 'gini' is the chosen impurity measure to calculate the highest information gain at each node for dividing the DT. In case of document data, the conditions depend upon the particular term in a text document of the news article. The data is divided repetitively until the leaf node cannot be further divided acquiring least information on them. The majority count of labels in the leaf node are used for classifying the textual data.

5.6 Ensemble methods

Such methods help to build a learning algorithm by combining the estimators to get robust classifier over single classifier. The boosting methods implemented on our dataset are Gradient Descent Boosting (GDB) and AdaBoost classifier for binary classification. The AdaBoost classifier assigns more weight to feature vectors which is difficult to handle and less weight to the features which can be easily handled. This process is repeated until the classifier correctly classifies the training data. The GDB model works on three elements such as weak learner, loss function and additive model as discussed in [16].

The other type of classifier that can be useful for balancing individual weakness of ML models is Voting classifier. In this paper, the Voting classifier has predicted the categories based on 'hard' voting which classifies the sample based on majority class label. To evaluate the classifiers discussed above, the performance metrics are defined and corresponding experimental results are discussed in the next section.

6 Evaluation Phase

The performance measures for binary classifiers applied in this paper has been evaluated with the help of a confusion matrix defined by four cells as shown in Table 5, where:

- cell 'a' counts the predicted document as 'Fake' when actually it is 'Fake', known as true positive (TP) rate.
- cell 'b' counts the predicted document as 'Real' when actually it is 'Fake', known as false positive (FP) rate.
- cell 'c' counts the predicted document as 'Fake' when actually it is 'Real', known as false negative (FN) rate.
- cell 'd' counts the predicted document as 'Real' when actually it is 'Real', known as true negative (TN) rate.

Table 5: A confusion matrix representation

Actual↓ F	$\mathbf{Predicted} ightarrow$	Fake	\mathbf{Real}
Fake		TP(a)	FP (b)
Real		FN(c)	TN(d)

The conventional performance measure has been evaluated from the above confusion matrix cells. The measures computed from the matrix are precision as represented by Eq. (9), recall by Eq. (10), specificity by Eq. (11), accuracy by Eq. (12), error by Eq. (13), F1-score by Eq. (14) as shown below.

$$Precision \ (Pr) = \frac{a}{a+b} \tag{9}$$

$$Recall (Re) = \frac{a}{a+c} \tag{10}$$

$$Specificity \ (Sp) = \frac{d}{d+b} \tag{11}$$

Accuracy
$$(Acc) = \frac{a+d}{n}$$
, where $n = a+b+c+d > 0$

(12)

$$Error (Err) = \frac{b+c}{n}, where \ n = a+b+c+d > 0 \ (13)$$

$$F1 - score \ (F1) = \frac{2 \times Pr \times Re}{Re + Pr}$$
(14)

The predictions made by the classification models are evaluated in this phase based on their performance metrics. The most intuitive performance measure is accuracy, which helps to predict the best model. Various machine learning models used in experiment are MultinomialNB (C1), Passive Aggressive (C2), Stochastic Gradient Descent (C3), Logistic Regression (C4), SVC (C5), NuSVC (C6), LinearSVC (C7), Multi-Layer Perceptron (C8), Decision Tree (C9), AdaBoost (C10), Gradient Descent (C11), Voting (C12) and Multi-level Voting (C13) classifiers. To evaluate these models, a comparative analysis has been shown in Figure 11. The experiment is performed on three (News Trends, Kaggle, Reuters) different corpus. In this paper, Tf-Idf, CV, and HV feature extraction techniques are used to extract the feature vectors from the documents of the chosen corpus. In Table 6, the accuracy measure of various ML classifiers is compared.

The best accuracy retrieved by top 3 models in all three corpora is Linear Support Vector Classifier (LSVC), Passive Aggressive (PA) and Logistic Regression (LR) classifiers.

Other parameters used to evaluate the performance measure of classifiers used in this paper are precision, recall and F1-score. The precision metric helps to calculate the proportion of news article that are predicted fake and actually also belongs to the fake news article category. The comparative analysis of precision metric is shown in Table 7.

The recall metric helps to calculate the proportion of news article that are predicted fake but actually belongs to both fake and real articles. The comparative analysis of recall metric is shown in Table 8. The specificity metric helps to calculate the proportion of news article that are correctly predicted as a real news article known not to be fake. Specificity is measured as inverse of recall metric. The comparative analysis of specificity metric is shown in Table 9. Accuracy is only measured as a strong metric when both false negative and false positive values are closer to each other else the metric is not considered as a good performance measure. To convey a balance between recall and precision, F1score performance metric is selected to retrieve the best model. F1-score metric helps to take both false positives and false negatives into account. The comparative analysis of F1-score metric is shown in Table 10.

The objective of ROC is to notice the increase in false positive rates (FPR) with an increase in true positive rates (TPR) with a varying threshold of the classifiers used in this paper. The performance of class models at various thresholds is shown through graphs in Figure 12. The curve drawn in the graph is known as Receiver Operating Characteristic (ROC) curve. The ROC for News trends, Kaggle and Reuters are plotted using two parameters such as FPR and TPR as given by Eq. (15) and Eq. (16).

$$TPR = \frac{a}{a+c} \tag{15}$$

$$FPR = \frac{b}{b+d} \tag{16}$$

Here, a, b, c, and d represents the TP, FP, FN and TN rates, respectively. predicted document as 'Real' when actually it is 'Fake', known as false positive (FP) rate, c counts the predicted document as 'Fake' when actually it is 'Real', known as false negative (FN) rate, d counts the predicted document as 'Real' when actually it is 'Real', known as true negative (TN) rate.

6.1 Major findings

The major findings deal with the question of whether a trained classifier using old news articles can give accurate and efficient results to categorize the differences between fake and real content. It has been observed that the performance of classifying news article depends on the corpus and type of classification model. In our experiment, three corpora have been collected from three different sources [39] [40] [41]. Each corpus is divided into training (0.67) and testing (0.33) sets. The experiment was performed on these chosen datasets using Term frequency-Inverse document frequency (Tf-Idf), Count-Vectorizer (CV) and Hashing-Vectorizer (HV) feature extraction techniques. From the accuracy perspective, Passive Aggressive (93.2%) and LinearSVC (93.2%) outperform other models for all three (News Trends, Kaggle, Reuters) corpora. Whereas the Passive Aggressive (96%) and the LinearSVC (95.9%) performs best using Tf-Idf, Logistic Regression (94.9%) and Stochastic Gradient Descent (94.2%) performs best using CV, LinearSVC (90.6%) and Stochastic Gradient Descent (90.5%) performs best using HV individually for all three corpora.

A classifier is considered usable only if it achieves both high precision and recall. To average out the results of both precision and recall, F1-score is taken into consideration. On evaluating F1-score metric, it was observed that Passive Aggressive (93.3%), Stochastic



Fig. 11: Performance analysis using accuracy metric for (a) News Trends, (b) Kaggle and (c) Reuters dataset on basis of Tf-Idf, Hashing-Vectorizer and Count-Vectorizer feature extraction techniques

Table 6: Comparative analysis of accuracy measure using Machine Learning classifiers

MODELS	New	s Tren	ds	K	aggle		Reuters		
MODELS	Tf-Idf	\mathbf{CV}	HV	Tf-Idf	\mathbf{CV}	HV	Tf-Idf	\mathbf{CV}	HV
Multinomial Naïve Bayes	85.7	89.3	83.6	93.2	95.4	89	84.8	89.9	86.2
Passive Aggressive	93.5	89.4	86.6	98.3	97.6	95.7	96.2	94.3	88
Stochastic Gradient Descent	93.4	90.7	86.8	98	97.1	95.5	96.2	94.8	89.4
Logistic Regression	91.4	91.0	87.0	96.4	98	93.6	94.9	95.7	89.6
Support Vector Classifier	48.2	74.1	48.2	52.9	74	52.9	52.5	74.3	81.3
NuSVC	80.1	83.5	86.3	92.2	86.1	92.6	70	86.6	88.3
LinearSVC	93.6	87.9	87.3	97.9	97.6	95.3	96.3	94.4	89.3
Multi-Layer Perceptron	93	91.5	84.7	97.1	97.3	94.7	96.2	96.2	90.1
Decision Tree	81.3	80.4	75.2	95.6	96.2	91.3	87.7	88.5	80.6
AdaBoost	86.7	85.1	82.6	97.3	96.6	92.9	92.5	92.4	85.9
Gradient Boosting	89.2	88.6	85.3	98.5	98.0	95.7	92.3	92.5	87.6
Voting Classifier	93.8	92.1	87.3	98.3	98.3	95.9	96.1	96.4	90.3

Table 7: Comparative analysis of precision metric using Machine Learning classifiers

MODELS	New	s Tren	ds	K	aggle		Reuters		
MODELS	Tf-Idf	\mathbf{CV}	HV	Tf-Idf	\mathbf{CV}	\mathbf{HV}	Tf-Idf	\mathbf{CV}	HV
Multinomial Naïve Bayes	73.3	85.8	88.4	92.3	95.8	84.4	99.1	97.3	89.3
Passive Aggressive	94.5	90.0	87.6	98.7	96.9	97.1	96.6	93.6	91.1
Stochastic Gradient Descent	94.8	89.4	88.1	98.7	97.8	95.8	96.4	94.2	90.0
Logistic Regression	95.4	94.0	89.8	95.4	97.9	94.2	94.7	95.2	89.1
Support Vector Classifier	100	96.6	100	100	97.2	100	100	53.1	82.0
NuSVC	94.5	96.3	90.4	97.5	97.9	91.6	41.9	77.8	85.7
LinearSVC	96.1	88.8	87.8	98.1	96.5	96.2	93.6	94.1	89.4
Multi-Layer Perceptron	93.6	93.1	84.0	97.2	96.5	96.2	92.1	94.3	89.9
Decision Tree	80.2	81.9	75.0	97.2	96.6	92.8	87.1	88.3	80.1
AdaBoost	89.9	88.7	83.6	97.2	96.6	94.1	92	90.6	84.9
Gradient Boosting	91.9	92.6	89.0	98.4	97.8	95.4	90.7	90.4	89.9
Voting Classifier	95.8	94.2	89.2	98.8	97.8	96.8	96.8	95.0	89.9

Gradient Descent (93.5%) and LinearSVC (93%) outperform other models on all three news article corpora

using Tf-Idf, CV and HV feature extraction techniques. The Passive Aggressive (95.9%), Stochastic Gradient

Table of comparative	analysis	01 100001	1 11100110	asing hie			, eleberiter.	<i>,</i>	
MODELS	New	s Tren	\mathbf{ds}	K	Laggle		Reuters		
MODELS	Tf-Idf	\mathbf{CV}	HV	Tf-Idf	\mathbf{CV}	HV	Tf-Idf	\mathbf{CV}	HV
Multinomial Naïve Bayes	95.9	91.5	79.7	94.6	95.4	94	77.9	85.4	85.0
Passive Aggressive	92.2	88.1	85.0	98.0	98.3	94.8	96.0	95.3	86.7
Stochastic Gradient Descent	91.7	90.9	84.9	97.5	96.7	95.6	96.2	95.7	89.7
Logistic Regression	87.7	88	84.2	97.7	98.5	93.7	95.5	96.5	90.7
Support Vector Classifier	48.2	65.7	48.2	52.9	67.4	52.9	52.5	96.3	82.3
NuSVC	72.4	75.9	82.6	88.7	80.2	94.2	99.4	95.8	91.4
LinearSVC	91.1	86.4	86	97.8	98.9	94.9	96.6	95.1	90.1
Multi-Layer Perceptron	91.9	89.9	84.1	97.2	98.2	93.8	96.4	95.8	89.8
Decision Tree	80.7	78.3	73.9	94.5	96.1	90.9	89.2	89.6	82.3
AdaBoost	83.5	81.8	80.9	97.5	96.8	92.6	93.6	94.5	87.8
Gradient Boosting	86.4	85.0	82.0	98.7	98.4	96.3	94.2	95.1	89.9
Voting Classifier	91.6	89.7	85.1	98.0	98.9	95.4	95.0	96.6	89.8

Table 8: Comparative analysis of recall metric using Machine Learning classifiers

Table 9: Comparative analysis of specificity metric using Machine Learning classifiers

MODELS	New	s Tren	\mathbf{ds}	K	Caggle		Reuters		
MODELS	Tf-Idf	\mathbf{CV}	\mathbf{HV}	Tf-Idf	\mathbf{CV}	\mathbf{HV}	Tf-Idf	\mathbf{CV}	\mathbf{HV}
Multinomial Naïve Bayes	79.6	87.5	88	91.6	95.2	84.3	98.7	96.4	87.4
Passive Aggressive	94.8	90.5	88.2	98.5	96.6	96.6	96.2	93.1	89.6
Stochastic Gradient Descent	95	90.4	88.6	98.5	97.5	95.3	96	93.7	88.9
Logistic Regression	95.3	94	89.9	94.9	97.4	93.4	94.2	94.8	88.2
Support Vector Classifier	0	94.4	0	0	95.1	0	0	65.3	80.2
NuSVC	92.9	95.4	90.2	96.9	96.5	90.9	61.3	79.6	85.2
LinearSVC	96.2	89.3	88.5	97.8	96.2	95.7	95.9	93.5	88.4
Multi-Layer Perceptron	93.9	93.4	85.1	96.9	96.1	95.6	94.3	93.2	86.1
Decision Tree	81.7	82.4	76.4	96.8	96.2	91.7	86.1	87.3	78.6
AdaBoost	89.9	88.6	84.2	96.9	96.2	93.2	91.3	90.1	83.9
Gradient Boosting	92	92.5	88.9	98.2	97.5	94.8	90.1	89.9	85.2
Voting Classifier	95.9	94.3	89.5	98.6	97.6	96.3	96.4	94.6	88.8

Table 10: Comparative analysis of F1-score metric using Machine Learning classifiers

MODELS	New	's Tren	\mathbf{ds}	k	Caggle		Reuters		
MODELS	Tf-Idf	\mathbf{CV}	\mathbf{HV}	Tf-Idf	\mathbf{CV}	\mathbf{HV}	Tf-Idf	\mathbf{CV}	\mathbf{HV}
Multinomial Naïve Bayes	86.9	89.4	83.6	93.4	95.5	88.9	87.2	90.9	87.0
Passive Aggressive	93.4	89.2	86.5	98.3	97.5	95.9	96.2	94.4	88.8
Stochastic Gradient Descent	93.3	90.6	86.7	98.0	97.2	95.6	96.2	94.9	89.8
Logistic Regression	91.3	90.9	86.8	96.5	98.1	93.9	95.0	95.8	89.8
Support Vector Classifier	65	77.4	65	69.1	79.6	69.1	68.8	68.4	82.1
NuSVC	81.3	84.5	86.2	92.8	88.1	92.8	58.9	85.8	88.4
LinearSVC	93.5	87.8	87.2	97.9	97.3	94.9	95	94.5	89.7
Multi-Layer Perceptron	92.8	91.6	84.5	97.2	97.3	94.9	93.2	94.1	88.7
Decision Tree	81.1	80.2	75.1	95.8	96.3	91.8	88.1	88.9	81.1
AdaBoost	86.5	85.0	82.5	97.3	96.6	93.3	92.7	92.5	86.3
Gradient Boosting	89.1	88.5	85.3	98.5	98.0	95.8	92.4	92.6	89.9
Voting Classifier	93.7	91.8	87.1	98.3	98.3	96	95.8	95.7	89.8

Descent (95.8%) and LinearSVC (95.4%) performs best using Tf-Idf, Logistic Regression (94.9%), Stochastic



Fig. 12: TPR vs FPR at different classification thresholds for (a) Count-vectorizer on News Trends, (b) Count-vectorizer on Kaggle, (c) Count-vectorizer on Reuters, (d) TF-IDF on News Trends, (e) TF-IDF on Kaggle, (f) TF-IDF on Reuters, (g) Hash-vectorizer on News Trends, (h) Hash-vectorizer on Kaggle, (i) Hash-vectorizer on Reuters datasets

Gradient Descent (94.2%), Passive Aggressive (93.7%), LinearSVC(93.2) performs best using CV and Passive Aggressive (90.4%), Stochastic Gradient Descent (90.7%) and LinearSVC (90.6%) performs best using HV. The proposed multi-level voting model outperforms the Passive Aggressive model by 0.8%, 0.6%, 1.0% using Tf-Idf approach; outperforms the Logistic Regression by 2.6%, 0.7%, 0.8% using CV approach; outperforms the LinearSVC by 0.0%, 0.5%, 0.9% using HV approach on News Trends, Kaggle and Reuters corpus, respectively.

To evaluate the predictive performance of our approach, ROC_AUC is plotted. Passive Aggressive, Stochastic Gradient Descent and LinearSVC, Gradient Boosting, Logistic Regression outperform other ML models on basis of ROC_AUC metric with TPR > 0.97. Passive Aggressive, Stochastic Gradient Descent and LinearSVC gives TPR > 0.97 using Tf-Idf, Logistic Regression gives TPR > 0.98 using CV, Stochastic Gradient Descent, LinearSVC, Logistic Regression gives TPR >0.95 for News Trends, Kaggle and Reuters dataset, respectively. Based on the training time required by various ML classifiers, it has been observed that there is a trade off between efficiency and accuracy. The training time required by HV is less than Tf-Idf and CV technique but it compromises accuracy metric. It has been analyzed that the hashing technique is useful when the focus is to achieve high efficiency on a huge dataset. The two ML classifiers such as Logistic Regression and LinearSVC are chosen among other models which results in both high accuracy and efficiency to overcome the trade-off issue.

7 Proposed multi-level voting model

The proposed multi-level voting model not only helps to improve the accuracy but also helps to reduce the training time of the classifiers. The reduction in training time helps to increase the efficiency of our model by introducing parallel methods where the base learners are generated parallelLy. The motivation behind the proposed model is to analyze the independence between the base learners. Three levels are proposed to perform the experiment as discussed below.

Level 1: Sets of three ML classifiers are merged based on their performance metric (FP rate) to apply voting classifier. The voting models (VC1, VC2, VC3) are retrieved.

Level 2: A voting classifiers (VC4) is retrieved after merging the three models (VC1, VC2, VC3) on the basis of their false positive rate.

Level 3: The false predictions from voting classifier (VC4) are merged with PA and LinearSVC for Tf-Idf, LR and SGD for CV and SGD, LinearSVC for HV to get the final prediction.

On the basis of the minimum false positive (FP) rate, the ML models are merged to overcome the weakness of existing individual models. The minimum is the FP ratio, the more accurate the model will be to predict the content as fake. Three feature extraction techniques (Tf-Idf, CV, HV) are used to extract the features from a collected dataset. Based on the FP ratio, the models are selected and merged to give an appropriate prediction. First cluster at level 1, SGD (News

Trends), LR (Kaggle) and MLP (Reuters) using Tf-Idf; SGD (News Trends), LinearSVC (Kaggle) and MLP (Reuters) using CV; PA (News Trends), LR (Kaggle) and MLP (Reuters) using HV are merged to build the voting classifier (VC1). Second cluster at level 1, AdaBoost, Gradient Boosting and MultinomialNB (News Trends, Kaggle and Reuters) using Tf-Idf, CV and HV are merged to build VC2. Third cluster at level 1, SVC, NuSVC and DT (News Trends, Kaggle and Reuters) using Tf-Idf, CV and HV are merged to build VC3 as shown in Figure 13.

Based on FP ratio of three proposed voting classifiers (VC1, VC2, VC3), a fourth voting classifier (VC4) is retrieved at level 2 based on the FP rates of VC1, VC2 and VC3 classifiers. At level 3, PA, LinearSVC using Tf-Idf; SGD and LR using CV; SGD and LinearSVC using HV are retrieved based on TP rate and are further clustered with VC4. Based on FP rate, VC5 is retrieved to give the final prediction of the proposed model.

8 Evaluating the performance of multi-level voting model

In proposed multi-level voting classifier, the best three ML models are combined from each feature extraction technique as discussed in the previous section. It can be observed from Table 11 that the proposed model outperforms voting classifier by 0.73%, 0.66% and 0.13% using Tf-Idf, CV and HV, respectively in terms of accuracy metric. Similarly, the proposed model also gives significant improvement for precision, recall, specificity, and F1-score performance measures.

To compare the vectorization methods used in terms of training time required to train the data, Figure 14 depicts that the HV technique is more efficient than other feature extraction methods as the time required to train the proposed multi-level voting model is least for News Trends and Reuters dataset. To achieve better efficiency (less training time), Tf-Idf method is only suitable when data is not too huge whereas hashing can be used for huge datasets when there is a requirement to make a trade-off between efficiency and accuracy. After performing our experiment, some conclusions are drawn in next section which helps to analyze the best classifier to be chosen for both high efficiency and accuracy.

9 Comparison with Existing Works

The results given by our system are also compared with other existing works on fake news detection as shown



Fig. 13: Architecture of the proposed multi-level voting model

MODELS	New	s Tren	\mathbf{ds}	K	aggle		Re	euters		
MODELS	Tf-Idf	\mathbf{CV}	HV	Tf-Idf	\mathbf{CV}	HV	Tf-Idf	\mathbf{CV}	HV	
				Ac	curacy	7				
Voting classifier	93.8	92.1	87.3	98.3	98.3	95.9	96.1	96.4	90.3	
Multi-voting classifier	94.3	93.6	87.1	98.9	98.7	95.8	97.2	96.5	90.2	
				Pro	ecision	1				
Voting classifier	95.8	94.2	89.2	98.8	97.8	96.8	96.8	95	89.9	
Multi-voting classifier	96.4	94.3	89.6	99.1	98.3	96.8	98.4	97.6	90.4	
		Recall								
Voting classifier	91.6	89.7	85.1	98	98.9	95.4	95	96.6	89.8	
Multi-voting classifier	93.1	91.4	85.2	98.7	98.8	94.4	96.8	95.2	90.8	
				Spe	cificit	y				
Voting classifier	95.9	94.3	89.5	98.6	97.6	96.3	96.4	94.6	88.8	
Multi-voting classifier	96	92.4	86.1	98.2	97.1	93.6	95.7	94.6	87.9	
				$\mathbf{F1}$	-Score	;				
Voting classifier	93.7	91.8	87.1	98.3	98.3	96	95.8	95.7	89.8	
Multi-voting classifier	94.7	92.8	87.3	98.8	98.7	95.5	97.7	97	90.5	

Table 11: Comparative analysis of the proposed multi-level voting model with voting classifier

in Table 12, Table 13 and Table 14 using Reuters, Kaggle and News Trends corpora, respectively. Researchers have used ML approaches to perform fake news detection on various social media platforms. It has been analyzed from Table 12 that the feature extraction technique based on document frequency used by Mishu *et al.* (2016) for MultinomialNB (72%), SVC (78%) and voting classifier (89%) [45] does not give better accuracy than our proposed system for all three classifiers. In our proposed system, MultinomialNB (89.9%), SVC (81.3%), Voting (96.4%) outperforms Mishu *et al.* (2016) models.

It has been observed from Table 13 that the feature extraction technique based on Tf-Idf used by Ahmed *et al.* (2017) for ML classifiers does not gives better accuracy than our proposed system. The recorded observa-



Fig. 14: Training time comparison for (a) News Trends, (b) Kaggle and (c) Reuters dataset on basis of TF-IDF, Hash-vectorizer and Count-vectorizer feature extraction techniques

Authors	Proposed Approach	Features	Model Accuracy
Cai et al. (2008)	Presented a Bayesian classifica-	Topic-based and	LapPLSI - 74.6%
	tion approach using class-specific	document-based	
	features for automatic text clas-	modeling.	
	sification.		
Mishu <i>et al.</i> (2016)	Classified text document using various classifiers	Document frequency	MultinomialNB- 72%
			LR- 73.5%
			SGD- 76%
			SVC- 78%
			LinearSVC- 83.3%
			Voting- 89%
Analysis based on in- dividual ML models	Classified articles as fake or real using various classifiers.	TF-IDF	SGD- 96.2%
			PA- 96.2%
			LinearSVC- 96.3%
			MLP- 96.2%
			AdaBoost- 92.5%
			Voting- 96.4%
		Count-vectorizer	MLP- 96.2%
			DT- 88.5%
			LR- 95.7%
			$\rm Multinomail NB\text{-}89.9\%$
			Gradient Boosting- 92.5%
			SVC- 81.3%
		Hashing-vectorizer	NuSVC- 88.3%
Proposed Multi-level voting model	Fake news detection system is	TF-IDF	Multi-level voting- 97.2%
	proposed to achieve high accu-	Count-vectorizer	Multi-level voting- 96.5%
	racy and high efficiency.	Hashing-vectorizer	Multi-level voting- 90.2%

Table 12: Comparison of existing models with a proposed system for Reuters corpus

tion shows that our proposed multi-level voting model (97.2%) outperforms Gradient Boosting by 0.4% using Tf-Idf, LR by 2.3% using CV and NuSVC by 3.6% using HV feature extraction techniques, respectively for Kaggle corpus. From Table 14, it can be analyzed that our proposed multi-level voting model outperforms PA by 0.8% using Tf-Idf, MultinomialNB by 4.3% using

CV and NuSVC by 0.8% using HV feature extraction techniques, respectively for News Trends corpus. From the comparative analysis, it has been analyzed that our proposed multi-level voting model using all three feature extraction technique outperforms when compared to the individual performance metrics of ML models used by Ahmed *et al.* (2017) and Mishu *et al.* (2016)

Authors	Proposed Approach	Features	Model Accuracy
Ahmed <i>et al.</i> (2017)	Proposed a fake news detection model using n-gram analysis and ML techniques	n-gram based TF and TF-IDF	LSVM-92% KNN- 83.1% SVM- 86% DT- 89% SGD- 89% LR- 89%
Analysis based on indi- vidual ML models Proposed Multi-level voting model	Classified articles as fake or real using various classifiers.	Count-vectorizer	MultinomailNB- 93.2% SVC- 52.9% LR- 96.4% MLP- 97.1% DT- 95.6% Voting- 98.3%
		TF-IDF	LinearSVC- 97.9% SGD- 98% PA- 98.3% AdaBoost- 97.3% Gradient Boosting- 98.5% Voting- 98.3%
		Hashing-vectorizer	NuSVC- 92.2%
	proposed to achieve high accu-	Count_vectorizer	Multi-level voting- 98.9%
	racy and high efficiency.	Hashing-vectorizer	Multi-level voting- 95.8%

Table 14: Comparison of existing models with a proposed system for News Trends corpus

Authors	Proposed Approach	Features	Model Accuracy
Kuleshov et al. (2018)	Author shows the existence of	n-gram	NB- 93%
	adversarial examples in natural		
	language classification		
Analysis based on indi- vidual ML models	Classified articles as fake or real using various classifiers.	TF-IDF	PA- 93.5%
			SGD- 93.4%
			LinearSVC- 93.6%
			MLP- 93%
			DT- 81.3%
			AdaBoost- 86.7%
			Gradient Boosting- 89.2%
			Voting- 93.8%
			LR- 91.4%
		Count-vectorizer	MultinomailNB-89.3%
			SVC- 74.1%
		Hash-vectorizer	NuSVC-86.3%
Proposed Multi-level voting model	Fake news detection system is	TF-IDF	Multi-level voting- 94.3%
	proposed to achieve high accu-	Count-vectorizer	Multi-level voting-93.6%
	racy and high efficiency.	Hash-vectorizer	Multi-level voting- 87.1%

for developing a system for automatic classification of news article features to label them as fake or real news article. It has been observed from Table 12 that the proposed model outplays the PA model by 0.9% using Tf-Idf, LR model by 0.8% using CV and NuSVC model by 1.9% using HV feature extraction techniques, respectively when compared with their performance metrics.

10 Conclusion and Future Scope

The goal in this paper is to analyze the best known supervised technique for detecting fake news. When dealing with the machine learning classifiers, the key question is not to find a learning classifier superior to others but rather to find the conditions under which particular model outperform others for a given problem. The set of attributes extracted from the corpus taken uses three feature extraction techniques (Tf-Idf, CV and HV) to feed the extracted feature vectors into the selected machine learning models. Some characteristics taken from datasets for learning task are categorical attributes, missing values, headlines of the article, the body of article and the publisher name. Various classifiers, Multinomial Naïve Bayes (MultinomialNB), Passive Aggressive (PA), Stochastic Gradient Descent (SGD), Logistic Regression (LR), Support Vector classifier (SVC), NuSVC, LinearSVC, Multi-Layer Perceptron (MLP), Decision Tree (DT), AdaBoost, Gradient Boosting and Voting classifiers were analyzed on basis of performance measures. After analysing the classifiers, the focus was to utilize the strengths of one model to complement the weakness of another. So, the multilevel voting model was proposed, which integrates various ML models based on their FP rates to retrieve a news voting classifier to retrieve better prediction analysis. The developed model helps to solve the trade-off issue between accuracy and efficiency.

In future, a web-based GUI will be created for the proposed fake news detection system to classify the news as fake or real on real-time social media platforms such as Facebook, Instagram, Twitter, WhatsApp, *etc.* Also, the annotated dataset in form of images (with textual content written on them) will be collected and maintained from Facebook and Reddit platforms. The annotated dataset can be used for detecting fake images in future as no such dataset is available at present. The proposed system has the potential to provide an impulse to various emerging applications such as controlling the spread of fake news during elections, terrorism, natural calamities, crimes for the betterment of the society. Acknowledgements This Publication is an outcome of the R&D work undertaken in the project under the Visvesvaraya PhD Scheme of Ministry of Electronics & Information Technology, Government of India, being implemented by Digital India Corporation (formerly Media Lab Asia).

References

- N.J. Conroy, V.L. Rubin, Y. Chen, Proceedings of the Association for Information Science and Technology 52(1), 1 (2015)
- Fake news on whatsapp. http://bit.ly/2miuv9j. Last accessed: 27-08-2019
- 3. K.R. Canini, B. Suh, P.L. Pirolli, in <u>Privacy</u>, <u>Security</u>, Risk and Trust (PASSAT) and 2011 IEEE <u>Third Inernational Conference on Social Computing</u> (SocialCom), 2011 IEEE Third International Conference on (IEEE, 2011), pp. 1–8
- H. Ahmed, I. Traore, S. Saad, in <u>International Conference</u> on Intelligent, Secure, and Dependable Systems in <u>Distributed and Cloud Environments</u> (Springer, 2017), pp. 127–138
- 5. A. Gupta, H. Lamba, P. Kumaraguru, in <u>eCrime</u> <u>Researchers Summit (eCRS), 2013</u> (IEEE, 2013), pp. 1– 12
- 6. A.H. Wang, in <u>Security and cryptography (SECRYPT)</u>, proceedings of the 2010 international conference on (IEEE, 2010), pp. 1–10
- F. Benevenuto, G. Magno, T. Rodrigues, V. Almeida, in Collaboration, electronic messaging, anti-abuse and spam conference (CEAS), vol. 6 (2010), vol. 6, p. 12
- I. Sen, A. Aggarwal, S. Mian, S. Singh, P. Kumaraguru, A. Datta, in Proceedings of the 10th ACM Conference on Web Science (ACM, 2018), pp. 205–209
- 9. K. Garimella, G. Tyson, arXiv preprint arXiv:1804.01473 (2018)
- J.A. Caetano, J.F. de Oliveira, H.S. Lima, H.T. Marques-Neto, G. Magno, W. Meira Jr, V.A. Almeida, arXiv preprint arXiv:1804.00397 (2018)
- T. Pontes, M. Vasconcelos, J. Almeida, P. Kumaraguru, V. Almeida, in Proceedings of the 2012 ACM conference on ubiquitous computing (ACM, 2012), pp. 898–905
- P. Dewan, M. Gupta, K. Goyal, P. Kumaraguru, in Proceedings of the 5th IBM Collaborative Academia Research Exchange Workshop (ACM, 2013), p. 6
- M. Weimer, I. Gurevych, M. Mühlhäuser, in Proceedings of the 45th Annual Meeting of the ACL on Interactive Poster and Demonstration Sessions (Association for Computational Linguistics, 2007), pp. 125–128
- 14. W. Wei, X. Wan, arXiv preprint arXiv:1705.06031 (2017)
- Y. Chen, N.J. Conroy, V.L. Rubin, in <u>Proceedings of</u> the 2015 ACM on Workshop on Multimodal Deception Detection (ACM, 2015), pp. 15–19
- K. Shu, A. Sliva, S. Wang, J. Tang, H. Liu, ACM SIGKDD Explorations Newsletter 19(1), 22 (2017)
- V. Rubin, N. Conroy, Y. Chen, S. Cornwell, in Proceedings of the Second Workshop on Computational Approaches to Deception Detection (2016), pp. 7–17
- P. Dewan, P. Kumaraguru, Social Network Analysis and Mining 7(1), 15 (2017)
- P. Jain, P. Kumaraguru, in Proceedings of the 3rd IKDD Conference on Data Science, 2016 (ACM, 2016), p. 6
- 20. B. Markines, C. Cattuto, F. Menczer, in <u>Proceedings</u> of the 5th International Workshop on Adversarial <u>Information Retrieval on the Web</u> (ACM, 2009), pp. 41– 48

- F. Benevenuto, T. Rodrigues, V. Almeida, J. Almeida, M. Gonçalves, in Proceedings of the 32nd international ACM SIGIR conference on Research and development in information retrieval (ACM, 2009), pp. 620–627
- 22. V. Qazvinian, E. Rosengren, D.R. Radev, Q. Mei, in Proceedings of the Conference on Empirical Methods in Natural Language Processing (Association for Computational Linguistics, 2011), pp. 1589–1599
- 23. V.L. Rubin, N.J. Conroy, Y. Chen, in <u>Hawaii</u> International Conference on System Sciences (2015)
- 24. A. Magdy, N. Wanas, in <u>Proceedings of the 2nd</u> international workshop on <u>Search and mining</u> user-generated contents (ACM, 2010), pp. 103–110
- 25. S. Volkova, K. Shaffer, J.Y. Jang, N. Hodas, in Proceedings of the 55th Annual Meeting of the Association for Computational Linguistics (Volume 2: Short Papers), vol. 2 (2017), vol. 2, pp. 647–653
- 26. S. Chhabra, A. Aggarwal, F. Benevenuto, P. Kumaraguru, in <u>Proceedings of the 8th Annual</u> <u>Collaboration, Electronic messaging, Anti-Abuse</u> <u>and Spam Conference</u> (ACM, 2011), pp. 92–101
- 27. A. Aggarwal, A. Rajadesingan, P. Kumaraguru, in <u>eCrime Researchers Summit (eCrime), 2012</u> (IEEE, 2012), pp. 1–12
- G. D'Angelo, F. Palmieri, S. Rampone, Information Sciences 486, 31 (2019)
- Y. LeCun, Y. Bengio, G. Hinton, nature **521**(7553), 436 (2015)
- 30. N. Ruchansky, S. Seo, Y. Liu, in <u>Proceedings of the</u> 2017 ACM on Conference on Information and Knowledge <u>Management</u> (ACM, 2017), pp. 797–806
- 31. F. Ahmed, M. Abulaish, in <u>Trust, Security and Privacy</u> in Computing and Communications (TrustCom), 2012 <u>IEEE 11th International Conference on</u> (IEEE, 2012), pp. 602–608
- S.M. SIRAJUDEEN, N.F.A. AZMI, A.I. ABUBAKAR, Journal of Theoretical & Applied Information Technology 95(17) (2017)
- 33. H. Gao, J. Hu, C. Wilson, Z. Li, Y. Chen, B.Y. Zhao, in Proceedings of the 10th ACM SIGCOMM conference on Internet measurement (ACM, 2010), pp. 35–47
- 34. P. Dewan, P. Kumaraguru, in Privacy, Security and Trust (PST), 2015 13th Annual Conference on (IEEE, 2015), pp. 85–92
- A. Aggarwal, S. Kumar, K. Bhargava, P. Kumaraguru, (2018)
- 36. A. Gupta, P. Kumaraguru, Twitter explodes with activity in mumbai blasts! a lifeline or an unmonitored daemon in the lurking? Tech. rep. (2012)
- 37. A. Gupta, H. Lamba, P. Kumaraguru, A. Joshi, in Proceedings of the 22nd international conference on World Wide Web (ACM, 2013), pp. 729–736
- 38. A. Gupta, P. Kumaraguru, in <u>Proceedings of the 1st</u> workshop on privacy and security in online social media (ACM, 2012), p. 2
- News trends database. https://bit.ly/2zVRLxK. Last accessed: 18-10-2017
- Kaggle database. https://bit.ly/2Ex5VsX. Last accessed: 24-10-2017
- 41. Kaggle database. https://bit.ly/2BmqBQE. Last accessed: 22-10-2017
- 42. T. Pontes, G. Magno, M. Vasconcelos, A. Gupta, J. Almeida, P. Kumaraguru, V. Almeida, in <u>Data Mining</u> <u>Workshops (ICDMW), 2012 IEEE 12th International</u> <u>Conference on (IEEE, 2012), pp. 571–578</u>
- 43. O. Batchelor, Reference Services Review 45(2), 143 (2017)

- 44. A. Alahmadi, A. Joorabchi, A.E. Mahdi, in <u>2013</u> 7th <u>IEEE GCC Conference and Exhibition (GCC)</u> (IEEE, 2013), pp. 108–113
- S.Z. Mishu, S. Rafiuddin, in <u>Computer and Information</u> <u>Technology (ICCIT), 2016 19th International Conference</u> on (IEEE, 2016), pp. 409–413